The impact of guided inquiry-based learning in enhancing problem-solving skills among Grade 11 Development Studies learners: Insights from one high school in Maseru district



THE NATIONAL UNIVERSITY OF LESOTHO

A thesis submitted in fulfilment of the requirement for the award of degree of Master of Arts in Education (MA.Ed) in the Faculty of Education, Department of Language and Social Education (LASED)

By

PUSELETSO MATHAHA Under the supervision of

PROF. RASELIMO M. G.

SEPTEMBER 2024

Declaration statement

I Puseletso Mathaha (201701827) hereby announce that this dissertation entitled "*The impact of guided inquiry-based learning in enhancing problem-solving skills among grade 11 Development Studies learners: Insights from one high school in Maseru district*" is my original work and has not been published or submitted previously for any personal or academic purpose. I declare that the materials and sources used in this study have been duly acknowledged in-text accompanied by a compiled list of references.

Signature:



Ethics statement

I declare that in conducting this study, I observed all necessary ethics indicated by The National University of Lesotho. The human participants were involved in the current investigation. Therefore, ethical considerations including informed consent and confidentiality were adhered to.

Certification

Having met all the necessary requirements of the university, this certifies that this thesis deserves an award of the Master of Arts in Education under the Faculty of Education.

Supervisor: PROF. RASELIMO M. G.___

Head of Department: DR. MAKUMANE M.

Dean Faculty of Education: PROF MOSIA P. A.

Supervisors' statement

I **Mohaeka Raselimo** affirm that I have approved the submission of this thesis which was under my supervision.

no. Signature

Dedication

I dedicate this thesis to my beloved daughter, Kananelo Alice. Dear daughter, you have been my greatest source of inspiration and strength through every uncertainty and challenge we faced together. This is not just an academic achievement, as it may appear, but a testament to our journey together—a journey marked by love, perseverance, and unwavering support. To myself, this work represents not only my academic endeavours but also my commitment to self-improvement and resilience.

Acknowledgements

I would like to express my heartfelt gratitude to those who significantly supported me through this journey. First and foremost, I thank God Almighty for granting me strength, courage, and wisdom during this challenging yet rewarding experience. I sincerely acknowledge His mercy and guidance throughout.

To my supervisor, Prof. Raselimo, I extend my deepest appreciation for your support, constructive criticism, and guidance. Your mentorship was instrumental in shaping my work and positively impacted the completion of my thesis. I am indeed grateful for the time and sacrifices you made to attend to my work.

I also wish to thank my lecturers, Dr. Musa Adekunle and Dr. Tawanda Mataka, for instilling in me a sense of confidence and laying the foundation for this research. Each of you uniquely contributed to my academic experience in a positive way.

The learners and teachers from the school that I carried the intervention, your participation is highly appreciated.

To my siblings—Nthakoana, Lekhanooe, Moliehi, Mampale, and Mots'elisi—thank you for your encouragement and motivation during this journey. It was not an easy one, but with your continuous support, I enjoyed every moment of it.

I would also like to acknowledge my friend, Bokang Semoko, for the love and support you have shown me. You stood by me through late-night sessions to meet stressful deadlines. Your companionship made this journey not only successful but also memorable.

Lastly, I extend my sincere gratitude to all my classmates for their support and teamwork, which positively impacted my work. To Neo Selikane, Puseletso Lebajoa, and Rets'elisitsoe Lebona, your emotional support is deeply appreciated.

Abstract

Teaching and learning in the 21st century necessitate a paradigm shift in the educational system, with a stronger emphasis on learner-centered methods. Constructivist theorists posit that learnercentered approaches can foster the development of skills valuable in both academic and personal contexts. This study aimed to investigate the efficacy of guided inquiry-based learning (IBL) in enhancing problem-solving abilities among Grade 11 Development Studies learners at a high school in Maseru. The theoretical framework for this study was grounded in the IDEAL model of problem-solving, developed by Bransford and Stein (1984). A two-group pre-test and post-test quasi-experimental design was employed to collect quantitative data on the effectiveness of guided IBL compared to traditional teaching and learning methods, within a positivist paradigm. The findings indicate that guided IBL is more effective in improving problem-solving skills among Development Studies learners than traditional learning methods. Specifically, the study revealed that at the pre-test level, both the control and experimental groups were statistically equivalent, with a recorded p-value of 0.702 (>0.05), indicating no significant differences in the learners' initial problem-solving abilities. After the intervention, the experimental group significantly outperformed the control group (p = 0.000 < 0.05). Furthermore, observations showed that learners found it easier to solve problems when actively engaged in the learning process through inquiry. The Likert scale responses also indicated that learners were satisfied with the use of guided IBL in their classroom. While the study suggests that learners are better equipped to solve problems when actively engaged in the classroom, it also emphasises that teachers should facilitate this process by allowing students to ask questions and providing ongoing support throughout the learning process. Additionally, the study recommends that teachers be encouraged to adopt learner-centered approaches, such as guided IBL, to help learners acquire critical skills. Assessment and monitoring mechanisms should also be implemented to ensure the effective adoption of these teaching and learning strategies.

Keywords: Problem-solving, Education for Self-reliance, Guided IBL and 21st-century skills

Table of Contents

Declaration statement	i
Ethics statement	i
Certification	i
Supervisors' statement	ii
Dedication	ii
Acknowledgements	ii
Abstract	iv
List of Figures	viii
List of Tables	viii
CHAPTER 1	1
1.0 Introduction	1
1.1 Motivation	1
1.2 Background	2
1.3 Problem statement	5
1.4 Statement of Purpose	6
1.5 Research questions	7
1.5.1 Main research question	7
1.5.2 Sub-research questions	7
1.6 Significance of the study	7
1.7 Delimitation and scope of the study	7
1.8 Operational definition of key terms	8
1.9 Organisation of the Study	8
1.10 Chapter Summary	9
CHAPTER 2	
2.0 Literature review	
2.1 Theoretical underpinning	
2.2 Conceptual literature	
2.3 Inquiry-based learning	
2.3.1 Guided inquiry-based learning	
2.3.2 Benefits of IBL in teaching and learning	14
2.3.3 Phases of Inquiry-Based Learning	
2.3.4 Constraints of Implementing Inquiry-Based Learning in a Classroom	

2.3.5 How the Constraints of Implementing IBL in a Classroom Can Be Addressed	18
2.4 Problem-solving	19
2.5 Impact of traditionalised learning on learner's acquisition of skills	21
2.6 Empirical review	22
2.7 Gap and Appraisal	25
2.8 Conceptual framework	25
2.9 Chapter Summary	29
CHAPTER 3	30
3.0 Methodology	30
3.1 Introduction	30
3.2 Paradigm	30
3.3 Research approach	31
3.4 Research design	32
3.5 Population, Sample, and Sampling Techniques	33
3.6 The Choice of the School and Its Profile	34
3.7 Data Collection Instruments	35
3.8 Treatment procedure	36
STEP 1	38
STEP 2	38
STEP 3	38
STEP 4	39
STEP 5	40
3.9 Data analysis methods	40
3.9.1 Pre-test data analysis	41
3.9.2 Observation Checklist Analysis	42
3.9.4 Likert scale analysis	44
3.10 Assumptions that were followed before running the SPSS tests	44
3.11 Validity and Reliability	46
3.12 Ethical Considerations	47
3.14 Chapter Summary	47
Chapter 4	48
4.0 Results Presentation and Interpretation	48
4.3 The current level of problem-solving among learners in DS	50

4.4 The influence of guided inquiry-based Learning on learners' problem-solving skills in DS5	54
4.5 Post-test results presentation6	51
4.6 The level of satisfaction among learners in DS on the use of guided inquiry-based learning in teaching problem-solving skills.	55
4.7 Chapter summary6	;9
Chapter 5	0'
5.0 Discussion of the results, conclusion and recommendations7	0'
5.1 Introduction	0'
5.2 The current level of problem-solving skills among learners in DS7	0'
5.3 The influence of guided inquiry-based learning in enhancing problem-solving skills of learners in DS7	1
5.4 The post-test results discussion7	′4
5.5 The level of satisfaction among learners on the use of guided inquiry-based learning in DS class	75
5.6 Conclusion7	7
5.7 Recommendations	'8
5.8 Limitations of the Study7	'9
5.9 Areas for Further Research	'9
5.10 Reflections	'9
REFERENCES	30
Appendices9) 5
Appendix 1 Observation checklist9) 5
Appendix 2 Likert scale9) 7
Appendix 3 Pre-test, Post-test	99
Appendix 4 Introductory letter)1
FACULTY OF EDUCATION)1
Appendix 5 Informed consent)2
Appendix 6 The language editors' letter10)4
P.O. Roma 180)4
Faculty of Education)4
Appendix 7 Turnitin Report10)3

List of Figures

Figure 2.1 The IDEAL model of problem-solving in teaching and learning	.11
Figure 2. 2 The implementation of IBL in a classroom	16
Figure 2. 3 The conceptual framework	23
Figure 3.1 Experimental group	33
Figure 3. 2 Control group	33
Figure 3.3 The steps engaged in teaching problem-solving using guided IBL	38
Figure 3. 4 Example of the pictures that were presented to learners	39
Figure 3. 5 An illustration of how the analysis was carried out	43
Figure 3. 6 Variable view of dataset loaded in SPSS	44
Figure 4.1 An illustration of learners' responses on a given activity	56
Figure 4. 2 An Observation Checklist for the last lesson of the intervention	59
Figure 4. 3 Learners' exercise on problem-solving	60

List of Tables

Table 2.1 Bloom's Taxonomy of Verbs	22
Table 3.1 The normality test	.45
Table 4.1 Profile for Experimental group	.49
Table 4.2 Profile for the Control group	.49
Table 4.3 The mark scores of learners at the pre-test level	.50
Table 4. 4 The mean score of learners in DS who took a pre-test	51
Table 4.5 An independent sample t-test of the control and experimental groups before treatment.	52

Table 4.6 The mean scores of Control and Experimental groups before treatment	1
Table 4. 7 An observation checklist for the first lesson on guided inquiry-based learning54	1
Table 4.8 Results from observation checklist of the third lesson of the intervention	7
Table 4.9 The mean scores of learners after treatment	51
Table 4.10 The ANCOVA summary of the test scores of experimental and control groups6	52
Table 4.11 The ANCOVA summary of the test scores of experimental and control groups6	54

CHAPTER 1

1.0 Introduction

This chapter provides the background to the study, with a primary focus on Development Studies as a subject and the Inquiry-Based Learning approach. It also includes key components such as the problem statement, the significance of the study, the research questions, and the delimitation of the study.

1.1 Motivation

Problem-solving is one of the most crucial 21st-century skills, enabling learners to better understand their subjects (New Education Policy, 2020). In Development Studies (DS), problemsolving is recognised as a key ability that helps learners achieve the concept of education for selfreliance, a core principle that this subject aims to instill. However, during high school, both my classmates and I struggled to answer questions that required problem-solving skills. Our teachers often delivered content passively, which seemed to impede our ability to develop these essential skills.

Additionally, through my experience as a practical teacher, I observed that learners today continue to struggle with questions involving problem-solving. Both my former teachers and the colleagues I have worked with appear to find it challenging to devise effective strategies for helping learners develop these skills. One major contributing factor to this problem, in my view, is the method of content delivery. Problem-solving is inherently a learner-centered skill, and it is logical that it should be taught using learner-centered approaches.

These observations have motivated me to conduct this study, which aims to experiment with guided Inquiry-Based Learning (IBL) to assess its impact on enhancing problem-solving skills among learners in DS. I chose guided IBL over other teacher-centered approaches because it encourages learners to ask questions and explore ideas independently. My exposure to postgraduate studies has further fuelled my desire to make a meaningful contribution to the teaching and learning of DS.

The next section presents the background to the study.

1.2 Background

Development Studies (DS) as an academic discipline and school subject has existed since the World War I era when economic development was seen as a powerful tool for emancipation and overcoming societal backwardness (Currie-Alder, 2016). Following World War II, it became evident that economic development alone could not address the broader aspects of societal advancement. Consequently, the term "Development Studies" was introduced to assist post-colonial countries in their reconstruction efforts after the devastation caused by the war (Berma, 2004; Currie-Alder, 2016; Hettne, 1990; Thorbecke, 2006). This led to the introduction of DS as a school subject aimed at educating learners about various aspects of development and the strategies that can be employed for national reconstruction.

Various scholars have characterised DS as both multidisciplinary and interdisciplinary, with its aims deeply rooted in the educational system. Specifically, Arocena (2016), Hettne (1990), the Ministry of Education and Training (MoET, 2020), and Sumner (2006) view DS as a problem-based, interdisciplinary field that seeks to understand the social, political, environmental, and economic issues influencing the development process, particularly in developing countries. Among its primary objectives, DS aims to equip learners with analytical tools for understanding societal problems and to encourage their active participation in community development initiatives to address such challenges (MoET, 2020).

Several African countries introduced DS in their post-colonial education systems to eliminate colonial narratives and promote education for self-reliance. For instance, Komba et al. (2019) noted that the Tanzanian government integrated DS into secondary education to equip students with an understanding of the complexities surrounding development. Furthermore, Kontinent et al. (2022) affirm that DS was included in the Tanzanian curriculum to support the goal of education for self-reliance, a concept introduced by Julius Nyerere, Tanzania's first Prime Minister after independence.

Similarly, Botswana introduced DS around 1977 as part of its education for Kagisano (Boikhutso, 2013). This initiative aimed to create a home-grown school curriculum in junior and secondary grades that would address societal problems through education for self-reliance (Bagwasi, 2017; Boikhutso, 2013). Education for self-reliance equips learners with skills to tackle real-world

challenges within their communities (Komba et al., 2019). Bagwasi (2018) further observes that incorporating DS into Botswana's educational system was intended to maintain the relevance of education by addressing real-life issues affecting growth and development.

In Lesotho, the 1974 curriculum diversification reform included the introduction of DS. Raselimo and Mahao (2015) assert that practical disciplines such as home economics, agriculture, DS, and technological studies were added to the curriculum to signify a shift from conventional to more learner-centered teaching and learning. According to the Ministry of Education, Sports, and Culture (1982), these disciplines were also envisioned to attain the objectives of education for self-reliance through "education with production." Tanzanian philosopher Julius Nyerere viewes education for self-reliance as one that helps learners creatively apply knowledge and skills to solve societal problems (Schreiber-Barsch, 2023). Raselimo and Mahao (2015) emphasise that DS was designed to foster a shift toward learner-centered learning while promoting self-reliance among learners, preparing them for work environment and life's challenges.

Given the aforementioned perspectives on the introduction of DS in various countries, it is evident that this subject is inherently learner-centered. Its aim to promote education for self-reliance underscores the importance of problem-solving skills. Among the learner-centered techniques outlined by Sivakumar (2018), Inquiry-Based Learning (IBL) is identified as particularly effective for teaching social studies. Lesotho's education system emphasises engaging learners in inquiry processes to develop higher-order thinking skills (MoET, 2009, p.18; MoET, 2021). Furthermore, Lekhanya (2020) describes DS pedagogy as encompassing problem- or inquiry-based education, project-driven education, cooperation, problem-solving, and critical thinking.

IBL, emerging from the constructivist approach, is a learner-centered pedagogy that fosters highlevel thinking skills such as problem-solving, creative thinking, collaboration, and critical thinking (Rodriguez et al., 2019). IBL is categorised into three forms: structured, guided, and open inquiry. Orosz et al. (2023) define structured IBL as a method where teachers pose a question or issue for learners to investigate, with the expected answers already known. Guided IBL involves the teacher acting as a facilitator while learners explore a problem to find solutions, fostering autonomous thinking, problem-solving, and collaboration (Pappas, 2014). Open-ended IBL allows learners to pose their own questions and design their investigations from start to finish. Among the three types, guided IBL appears to be the most suitable for promoting problem-solving skills, which are essential for developing self-reliance. Palupi (2020) describes guided IBL as a strategy where learners are guided by the instructor through a structured inquiry framework, allowing them to take ownership of their learning by posing questions, conducting research, and drawing conclusions. More, Orosz et al. (2023) note that guided IBL enables learners to build knowledge through interaction and experience, actively engaging them in the inquiry process.

The guided IBL approach suggests that learners should be allowed to explore ideas and develop strategic measures for addressing life challenges. In the context of DS, Dambudzo (2015) highlights that DS covers topics related to sustainable development, where learners develop skills to maintain environmental sustainability. Other DS topics that engage learners in real-world issues include globalisation, population growth, foreign aid and investment, government, environmental sustainability, and sustainable development (MoET, 2020). Guided IBL can be employed to teach these topics, such as by asking learners to explore environmental sustainability issues and propose mitigation strategies.

Lesotho's Curriculum and Assessment Policy Statement (MoET, 2009) and the Basic Education Curriculum Policy (MoET, 2021) underscore the importance of IBL in primary and secondary education. The Curriculum and Assessment Policy (CAPs) specifically advocates for curriculum delivery that fosters creativity, independence, and survival skills, enabling learners to formulate and solve problems (MoET, 2009, p. 22). This policy reflects a formal endorsement of the relevance of IBL in Lesotho's educational system. Ralebese et al. (2022) also affirm that CAPs support IBL as an effective method for helping learners develop problem-solving skills and achieve self-reliance.

While IBL is recognised as important in Lesotho's curriculum implementation, Koekoe (2023) found that it is rarely used. Her research revealed that geography teachers have a positive perception of IBL but lack understanding of how to implement it in the classroom. This is attributed to a lack of clear guidelines on how to employ IBL in geography education. Orosz et al. (2023) highlight that the successful implementation of IBL in the classroom depends on teachers' willingness and understanding of how to guide learners through the inquiry process. Koekoe's

(2023) findings suggest that IBL can potentially be used to enhance problem-solving skills in DS education, contributing to education for self-reliance.

According to Ahuja (2021), problem-solving skills involve the cognitive processes of finding solutions and making informed decisions in specific situations. Learners must develop strong problem-solving abilities to be aware of local, national, and global challenges affecting development (MoET, 2018). This implies that learners should be taught critical strategies for addressing these challenges, such as devising solutions to poverty-related issues under the theme of poverty and development. Therefore, it is crucial to assist learners in developing problem-solving skills.

Lesotho's National Strategic Development Plan II (NSDP, 2018/19-2022/2023) advocates for education that emphasises equipping learners with the skills necessary to address both environmental sustainability and economic challenges. Similarly, the Lesotho Basic Education Curriculum Policy (MoET, 2021) emphasises that education should aim to provide students with the knowledge and skills required to succeed and compete in the global economy. This indicates that Lesotho's education system places significant importance on developing problem-solving abilities.

Numerous studies have documented the positive effects of IBL on learners' problem-solving abilities. For instance, Davrik et al. (2020) found that IBL enhances learners' creative thinking and problem-solving skills by allowing them to organise their knowledge systematically. Similarly, Gunawan et al. (2020), Hala and Xhomara (2022), and Sukontawaree et al. (2022) reported that IBL positively influences science teaching and learning, particularly by improving learners' attitudes toward science and their problem-solving abilities.

1.3 Problem statement

Problem-solving skills are critical in promoting education for self-reliance in Development Studies (DS), as this subject aims to equip learners with the creative abilities needed to address real-life challenges. However, the Examination Council of Lesotho (ECoL) 2019 markers' report indicated that DS learners struggle with questions requiring the application of problem-solving skills (ECoL, 2019). For instance, when asked to propose solutions to the effects of global warming on

biodiversity, learners were unable to provide adequate responses. This issue persisted in 2020, with learners failing to apply problem-solving skills and instead discussing adaptation strategies. The same problem was reported in 2022 and 2023, leading to poor performance in the subject. These recurring issues have been attributed to the ineffective instructional strategies employed in teaching DS.

Despite the Ministry of Education and Training (MoET, 2009) and the Lesotho Basic Education Curriculum Policy (LBECP, 2021) emphasising that the intended pedagogy in Lesotho's educational system should be learner-centered, with a strong focus on inquiry-based learning (IBL), DS teachers appear to rely heavily on rote learning. Lekhanya (2020) found that DS teachers frequently employ inappropriate pedagogical strategies, neglecting learner-centered approaches, including IBL. While Koekoe (2023) noted that IBL is rarely used due to teachers' lack of knowledge on its implementation, Lekhanya (2020) and Lekhanya and Raselimo (2022) did not report any significant use of IBL in DS instruction. Instead, they observed a tendency among DS teachers to prioritise rote learning and memorisation.

Given that IBL has been shown to enhance problem-solving and other 21st-century skills, it may be a suitable teaching approach for DS instruction. Although Koekoe (2023) conducted research on the application of IBL in Lesotho's geography education, there appears to be a lack of research or literature specifically addressing the use of IBL in DS education in Lesotho. In consequence, this study investigated the impact of guided IBL on learners' problem-solving abilities in the context of DS learning.

1.4 Statement of Purpose

The teaching and learning of Development Studies (DS) in Lesotho is intended to represent a transformation from teacher-centered to learner-centered approaches (Raselimo and Mahao, 2015). However, it appears that teachers often employ teaching strategies that do not align with the subject's aims. This study examined the potential of inquiry-based learning (IBL) in enhancing learners' problem-solving skills in DS.

1.5 Research questions

1.5.1 Main research question

What is the impact of guided inquiry-based learning in enhancing problem-solving skills among learners in DS?

1.5.2 Sub-research questions

What is the current level of problem-solving skills among learners in Development Studies (DS)?

What influence does guided inquiry-based learning (IBL) have on learners' problem-solving skills in DS?

What is the level of satisfaction among learners on the use of guided inquiry-based learning in DS class?

1.6 Significance of the study

This study is expected to be significant for teachers, learners, researchers, and the broader education system. Teachers may benefit as the study provides insights into the implementation of inquiry-based learning (IBL) in the classroom, offering a practical reference for incorporating guided IBL into their daily teaching practices. For learners, the study could serve as a resource for understanding how to engage in the inquiry process to effectively solve problems. The education system may find value in adopting components of this study to guide teachers in applying IBL in their instructional methods. Finally, other researchers might find the study's findings useful for identifying gaps in the existing literature and for conducting further research on IBL using different approaches to enhance their knowledge and understanding of this pedagogical strategy.

1.7 Delimitation and scope of the study

Bhosale (2024) defines delimitation and scope as the extent and boundaries of a study, which allow researchers to concentrate on specific aspects of their research. It is crucial for research to focus on essential data while avoiding irrelevant information. Clearly defining the scope and delimitation ensures that the research remains within its set parameters and achieves its objectives within the constraints of time, budget, and resources (Bhosale, 2024).

This study was limited to a single high school in Maseru that offers Development Studies as a subject, specifically focusing on grade 11 learners. Of the various 21st-century skills, the study

was delimited to examining learners' problem-solving skills in Development Studies and the use of guided Inquiry-Based Learning (IBL) as the instructional method.

1.8 Operational definition of key terms

This section defines key concepts as they pertain to the context of this study:

- **Problem-solving Skills**: Within the scope of this study, "problem-solving skills" refer to the abilities of learners to critically analyse a problem, explore pertinent information, and devise solutions to issues presented in questions requiring such skills.
- **Development Studies (DS) Subject**: Development Studies is a multidisciplinary and interdisciplinary subject aimed at equipping learners with an understanding of the political, economic, environmental, social, and cultural changes affecting development, particularly in developing countries (Sumner, 2006). In Lesotho, this subject seeks to provide learners with analytical tools to address real-world challenges and contribute to community development.
- **Impact**: In this study, "impact" denotes either a positive or negative effect or change resulting from the application of a particular teaching method.
- **Guided Inquiry-Based Learning (IBL)**: Guided IBL is a pedagogical approach derived from constructivist theories that enables learners to pose questions and engage in inquiry while receiving continuous support from the teacher.
- Education for Self-Reliance: This term refers to educational practices designed to equip learners with both academic and practical skills applicable not only in academic contexts but also in everyday life.
- **21st-Century Skills**: These skills encompass competencies that educational systems aim to develop in learners to promote independence and self-reliance, including problem-solving, critical thinking, and analytical skills.

1.9 Organisation of the Study

The current study is structured as follows: Chapter 1 introduces the background, problem specification, research questions, significance, and delimitations of the study. Chapter 2 covers the theoretical foundation, conceptual framework, and reviews relevant empirical literature to identify prior research pertinent to the study. Chapter 3 details the methodologies employed for

data collection and analysis. Finally, Chapter 4 presents the findings, offers recommendations, and provides an analysis along with concluding remarks.

1.10 Chapter Summary

This chapter has introduced and contextualised the problem under investigation, establishing a foundation for the study. It has elucidated the research questions the study aims to address and outlined the parameters of the investigation.

CHAPTER 2

2.0 Literature review

This section encompasses the theoretical framework, conceptualisation of key terms, empirical review, and conceptual framework. Sridhar (2020) highlights the significance of the literature review in research, noting that it aids in identifying the originality and relevance of the research problem and situates the current study within the context of existing research. Similarly, Mudavanhu (2017) asserts that a thorough literature review is crucial for understanding what has been accomplished and what gaps remain in relation to the current study. The following section provides the theoretical underpinning for the research.

2.1 Theoretical underpinning

According to Sarah (2022), a theoretical framework serves as a foundational review of an established theory, providing a roadmap for research. The current study is guided by Bransford and Stein's (1984) IDEAL model of problem-solving. This model was designed with the intention of helping educators teach critical thinking and problem-solving skills through its five components (Nickols, 2020). Given that Development Studies (DS) inherently involves problem-based learning, aimed at equipping learners with the skills and knowledge necessary to tackle real-world challenges (MoET, 2020), the IDEAL model emerges as a suitable theoretical framework for DS instruction. Supporting this choice, Sippl (2021) found that the IDEAL model facilitates the development of problem-solving strategies, promoting independent problem-solving among learners.

The acronym IDEAL represents the steps of Recognizing the problem, Establishing goals, Investigating solutions, Acting, and Looking and learning (see Figure 2.1). Sippl (2021) notes that, without structured guidance, teachers may struggle to cultivate problem-solving skills in students. As a result, the IDEAL model is selected as the optimal framework for implementing guided inquiry-based learning to enhance problem-solving skills in DS. Its straightforward acronym is easier for learners to remember and for teachers to apply (Pratiwi et al., 2021). Additionally, Annizar et al. (2020) highlight the IDEAL model as a crucial step-by-step procedure for effective problem-solving development in the classroom. Annizar (2018) further emphasises the model's role in enhancing higher-order thinking skills. Therefore, the IDEAL model provides a clear

procedural approach for employing guided inquiry to improve problem-solving skills among DS learners. Figure 2.1 illustrates the IDEAL model framework.



Figure 2.1 The IDEAL model of problem-solving in teaching and learning

In continuation, Figure 2.1 illustrates the first step in the problem-solving process according to Bransford and Stein (1984): identifying the problem. Setyadi and Triyanto (2019) emphasise that at this initial stage, it is crucial for learners to receive guidance from the teacher to articulate the problem in their own words. In the context of teaching problem-solving skills in Development Studies (DS), this stage involves learners identifying the problem from the provided information. Bransford and Stein (1984) stress the importance of creating a supportive learning environment where each learner feels valued and is encouraged to participate in problem identification (Gusau & Mohamad, 2020).

Following problem identification, the next stage is defining the goals. In this instance, learners, with the teacher's assistance, outline the objectives necessary to resolve the problem (Setyadi & Triyanto, 2019). Clear definition of these goals provides a structured pathway for finding solutions. For instance, in teaching about overpopulation in a DS class, learners would set objectives to guide their problem-solving efforts. Martin et al. (2024) and Wong et al. (2018) highlight that teacher support during this stage, including providing relevant learning materials, is essential for developing learners' problem-solving skills.

The subsequent step involves exploring solutions to the identified problem. Based on the established goals, learners brainstorm potential strategies for addressing the problem (Bransford & Stein, 1984; Sippl, 2021). Annizar et al. (2020) emphasise that during this stage, learners must

make informed decisions about various problem-solving strategies, ensuring that the chosen strategies align closely with the problem under investigation.

The fourth step, acting, requires implementing the selected solutions. At this stage, the strategies are reviewed and assessed to determine the most effective course of action (Pratiwi et al., 2021). Finally, the step of looking and learning involves reflecting on the entire problem-solving process and learning from the experiences gained (Gusau & Mohamad, 2020). This reflection enables learners to refine their problem-solving skills and apply their knowledge more effectively in future situations (Bransford & Stein, 1984).

2.2 Conceptual literature

This section presents a comprehensive overview of the concepts reviewed in the current study, starting with a literature review on Inquiry-Based Learning (IBL) and concluding with the conceptual framework.

2.3 Inquiry-based learning

Inquiry-Based Learning (IBL) is a pedagogical approach rooted in constructivist theory, which emphasises active learner engagement and the construction of new knowledge. Sarbah (2020) asserts that constructivism has given rise to various learner-centered approaches, including IBL. Constructivists argue that learners must be actively involved in their educational experiences to effectively build new knowledge (Olusegun, 2015). Central to constructivist theory are the concepts of assimilation and accommodation, which are pivotal as they describe how learners integrate new information with existing cognitive structures, thereby enhancing their mental representations (Golder, 2018; Sarbah, 2020). This dynamic process results in the advancement of learners' thinking and skills as they assimilate new concepts.

IBL, as an active pedagogical strategy, emphasises the construction of knowledge through social interactions (Chris, 2023; Pedaste et al., 2015; Wale et al., 2020). Lippmann (2022) highlights that IBL engages learners with complex questions and problems, encouraging exploration, investigation, and analysis to foster new understandings. Furthermore, IBL aligns with decolonial teaching strategies aimed at developing higher-order thinking and a comprehensive set of knowledge, skills, and attitudes (Fransinescu, 2018). This method is significant for teaching

Development Studies (DS), as it facilitates student involvement and the development of new competencies.

Consistent with constructivist principles, IBL builds upon learners' prior knowledge, which serves as a foundation for constructing new understanding (Mackenzie, 2016; Gholam, 2019). The approach also supports social interaction and continuous scaffolding, where learners collaborate, interact, discuss, and exchange ideas (Mackenzie, 2016). Vygotsky's (1978) principle of scaffolding emphasises the need for ongoing support throughout the learning process. Sarbah (2020) illustrates that IBL involves learners in posing questions and seeking answers, underscoring the necessity of providing continuous assistance to enable learners to reach well-informed conclusions during their inquiry.

2.3.1 Guided inquiry-based learning

Inquiry-Based Learning (IBL) encompasses three main strategies: structured inquiry, open-ended inquiry, and guided inquiry. In this study, guided IBL was chosen to enhance learners' problem-solving skills in Development Studies (DS). Guided IBL involves learners constructing knowledge through problem exploration, questioning, and analysis with the support of the teacher (Pedaste et al., 2015). The teacher's role in this approach is to facilitate the development of specific skills by presenting ideas and providing relevant learning materials to help learners achieve their goals (Orosz et al., 2023). For example, in DS, a teacher might focus on improving learners' problem-solving skills related to pollution by guiding them in investigating its causes and reduction strategies.

Guided inquiry closely aligns with the concept of "scaffolding," which Spadafora and Downes (2020) define as a process where a more knowledgeable individual provides ongoing support to help a novice learner achieve higher-level thinking. In support, Wood et al. (1976), prominent advocates of scaffolding, describe it as "a process that enables a child or a novice to solve a task or achieve a goal that would be beyond his unassisted efforts" (p. 90). In guided inquiry, the teacher offers a framework and continuous support during the stages of questioning and information retrieval.

The principles of scaffolding and guided inquiry are rooted in the work of constructivist theorists such as Jerome Bruner and Lev Vygotsky. Bruner (1978) emphasised that learners, as active participants, benefit from ongoing guidance while exploring concepts independently. Vygotsky (1978) similarly noted that continuous assistance encourages learners to ask questions and seek deeper understanding. Thus, guided inquiry facilitates a supportive learning environment where interaction, collaboration, and idea-sharing are integral to the learning process (Pedaste et al., 2015). Teachers can also leverage peer support, allowing more knowledgeable learners to assist their peers in information gathering and problem-solving. For instance, grouping learners to investigate questions or solve problems collectively can enhance their learning experience through collaboration.

2.3.2 Benefits of IBL in teaching and learning

As a product of constructivism, which is inherently learner-centered, Inquiry-Based Learning (IBL) is recognised for its potential to foster active and self-directed learning (Wale & Bishao, 2020). IBL, as a constructivist approach, positions learners as creators rather than mere consumers of knowledge (Andrini, 2016). In an IBL environment, learners are encouraged to construct their own understanding by asking questions, engaging with materials, and exploring topics to gather information (Canning & Masika, 2022). This active engagement allows learners to take ownership of their learning, promoting a deeper understanding of the subject matter. Along similar lines, Freire (1970) argues that traditional, passive learning methods are dehumanising and hinder learners' comprehension and skill development.

IBL also promotes critical thinking and problem-solving skills. Heick (2023) argues that openended questioning, coupled with continuous support, helps learners approach problems systematically and apply effective solutions. These problem-solving skills are crucial for both academic success and real-world applications (Gillies, 2023). In Development Studies (DS), IBL is particularly valuable as it encourages self-reliance among learners, addressing a key concern in teaching this subject. Thus, fostering problem-solving skills is essential in DS education.

Moreover, IBL fosters curiosity and exploration, cultivating lifelong learning habits. Through independent information-seeking and exploration, learners uncover important concepts and develop a passion for learning that extends beyond formal education (Marshall, 2016). Gillies

(2023) supports this by highlighting that IBL encourages innovation and creativity, enabling learners to think outside the box and develop creative problem-solving skills applicable to real-life issues. For instance, learners exposed to IBL in a DS classroom may effectively apply problem-solving skills to environmental sustainability challenges.

Another advantage of IBL is its capacity to enhance collaboration and teamwork among learners. Pedaste et al. (2015) emphasise that IBL fosters interaction as learners pose questions and seek answers collectively. This collaborative approach not only deepens their understanding but also develops essential skills for addressing real-life problems (Spadafora & Downes, 2020). In a collaborative DS classroom, more knowledgeable learners can assist their peers in grasping difficult concepts, facilitating a supportive learning environment (Wood et al., 1976). For example, peers may help each other understand challenging DS concepts, further enriching the learning experience.

2.3.3 Phases of Inquiry-Based Learning

To implement guided Inquiry-Based Learning (IBL) for enhancing learners' problem-solving skills in Development Studies (DS), the phases of IBL as outlined by Pedaste et al. (2015) were employed. The IDEAL model of problem-solving, with its five components, provided a framework for effectively applying these phases of inquiry. This model illuminated how teachers can systematically integrate the five phases of IBL to improve problem-solving capabilities among learners. See Figure 2.2 for an illustration of the phases of IBL.



Figure 2. 2 The implementation of IBL in a classroom

Inquiry-Based Learning (IBL) is a flexible approach that does not follow a linear or uniformly prescribed process and can thus vary in its application (Pedaste et al., 2015). In this study, the phases of IBL as summarised by Pedaste et al. (2015) were selected over other models such as those proposed by Marshall (2013) and Bybee (2006). According to Chris (2023), Pedaste et al. (2015) outline five key phases of IBL that are pertinent to guided inquiry-based classrooms.

The first phase is orientation, which involves the positioning and presentation of the topic by the teacher (Beyrow et al., 2019). This phase aims to help learners understand the topic and stimulate their curiosity, potentially using resources such as YouTube videos, books, cartoons, or guest speakers (Scholl, 2023). The IDEAL model emphasises that learners should be guided to identify the problem related to the lesson topic and be encouraged to articulate the problem in their own words (Pratiwi et al., 2021).

The second phase, known as conceptualisation or formulating the research question, involves assisting learners in developing an inquiry question—a central idea around which they will build an action plan (Pedaste et al., 2015). Kori (2021) posit that the action plan should outline the procedures to address the problem identified in the first phase, including resources and steps to be followed. As presented in the IDEAL model, this phase corresponds to defining goals, where learners are guided to establish goals that directly relate to the problem (Setyadi & Triyanto, 2019). This guidance is crucial for formulating relevant solutions.

The third phase, investigation, focuses on collecting evidence and information. During this phase, learners use the procedures developed in phase two to explore, research, experiment, observe, and gather data related to the topic (Ernst et al., 2017). The teacher's role is to facilitate this process, helping learners find answers and explore various methods of discovering new knowledge, such as through fieldwork and direct observation (Lippmann, 2022). For example, when studying pollution, learners might investigate their environment or participate in field trips to observe factors contributing to pollution.

The fourth phase is conclusion, where learners reflect on and compare their findings from the investigation phase. According to Chris (2023), this phase involves evaluating the strategies explored to determine the most effective solutions to the problem, as outlined by the IDEAL model

(Gusau & Mohamad, 2020). This phase is critical for assessing whether the inquiry question has been adequately addressed and for organising and presenting the results in a coherent manner (Wale & Bishao, 2020).

The final phase is discussion, which focuses on the implications of the findings and involves stimulating high-order thinking through questions posed by the teacher (Lippmann, 2022). This corresponds to the looking and learning stage in the IDEAL model, emphasising the importance of reflecting on the problem-solving process and learning from it to enhance future performance.

2.3.4 Constraints of Implementing Inquiry-Based Learning in a Classroom

Inquiry-Based Learning (IBL) is widely recognised as a prominent teaching and learning approach in the 21st century. However, several constraints impede its effective implementation in the classroom.

Firstly, time constraints are a significant challenge. Kuykendall (2022) identifies that IBL is often more time-consuming compared to traditional teaching methods, as it requires learners to formulate questions, delve deeper into topics, and conduct research. This poses difficulties in an overloaded curriculum with strict timelines, making it challenging for both teachers and learners to cover extensive content within limited time frames (Koekoe, 2023).

Secondly, learners' readiness to engage in inquiry-based classrooms presents another barrier. Kramer et al. (2022) observe that learners accustomed to teacher-centered environments often struggle to participate or interact with peers in an IBL setting. Such learners may find it overwhelming to function independently, explore ideas, ask questions, and seek information from diverse sources (Kuykendall, 2022).

A lack of teacher training on implementing IBL also hinders its application in the classroom. Koekoe (2023) highlights that while geography teachers may understand IBL theoretically, they often lack practical skills for effective classroom implementation. Teachers need a thorough understanding of their subject matter and pedagogical skills tailored to learner-centered approaches, especially those relevant to 21st-century education. Further, Koekoe (2023) reiterates that teachers frequently lack support from relevant stakeholders for the practical application of new teaching methods, including IBL.

Resource shortages also impact the successful implementation of IBL. In addition, Kuykendall (2022) argues that limited resources, such as computers, internet access, and textbooks, can inhibit effective inquiry-based learning. Inadequate resources may hinder both teachers and learners from engaging fully in the inquiry process (Mandina, 2024). For instance, in a Development Studies (DS) classroom, the absence of visual aids like videos can impede learners' ability to investigate topics like sustainable development, thereby limiting their capacity to ask questions and gather relevant information.

2.3.5 How the Constraints of Implementing IBL in a Classroom Can Be Addressed

While implementing Inquiry-Based Learning (IBL) presents several constraints, various strategies can be employed to address these challenges and achieve successful implementation.

Firstly, targeted teacher training by the responsible educational authorities is crucial. Pesqueira (2020) suggests that the ministry or educational bodies should facilitate professional development opportunities for teachers. Specifically, Koekoe (2023) emphasises the importance of workshops and seminars designed to equip teachers with the skills and strategies necessary for engaging learners effectively in inquiry-based processes. Furthermore, Heick (2023) advocates for collaborative efforts among teachers, where they can share techniques and best practices for creating successful inquiry-based classrooms. Collaborative lesson planning and integration of lesson content into the inquiry framework can enhance the overall effectiveness of IBL.

Time constraints, which are a significant challenge for IBL implementation, can be mitigated by aligning inquiry procedures with the syllabus's end-of-level objectives (Kuykendall, 2022). This requires careful planning to ensure that inquiry activities are designed to complement and reinforce subject content and desired skills. For instance, in a Development Studies (DS) classroom, the topic of environmental sustainability can be seamlessly integrated into the inquiry process, focusing on aspects of the content that align with inquiry-based learning objectives.

To address the issue of learners' readiness for inquiry-based approaches, particularly those accustomed to teacher-centered instruction, a gradual implementation strategy can be effective (Marshall et al., 2016). This approach involves introducing guided IBL incrementally, allowing learners to adapt gradually to the new methods. Additionally, integrating inquiry processes into existing classroom practices can facilitate smoother transitions (Gillies, 2023). For example, in DS education, incorporating field-based learning opportunities allows students to engage in inquiry by posing questions and exploring ideas in real-world contexts. This hands-on approach can help learners become more familiar with and receptive to the inquiry-based learning process.

2.4 Problem-solving

The 21st-century education system demands an approach that cultivates higher-order thinking skills, essential for learners to navigate and thrive in an ever-evolving world. According to Bloom (1969), and Nachiappan et al. (2018), the development of these skills is pivotal for enabling learners to manipulate information effectively and apply solutions to complex problems. Higher-order thinking skills encompass problem-solving, critical thinking, communication, creative thinking, and analytical thinking.

Nachiappan et al. (2018) emphasise that higher-order thinking extends beyond rote memorisation and reproduction of knowledge. It involves a deeper understanding of facts and ideas, enabling learners to evaluate and synthesise information. This underscores the need for educators to shift away from traditional teaching methods that do not promote learners as producers of knowledge. Instead, teachers should embrace 21st-century pedagogical practices and focus on skill development.

Problem-solving, a fundamental 21st-century skill, is highlighted by Rahman (2019) as a critical challenge in modern education. It involves cognitive processes aimed at achieving objectives through various strategies and techniques (Fuke, 2017). Adeoye et al. (2023) and Fissore et al. (2021) describe problem-solving as an intellectual process of exploring ideas to provide solutions to specific issues. The development of problem-solving skills is inherently learner-centered (Adeoye et al., 2023; Marshall, 2022), allowing learners to explore innovative solutions and engage in collaborative efforts to tackle problems (Rahman, 2019). In the same vein, Fuke (2017) asserts that problem-solving should be integrated into the content being taught, rather than being

presented as an abstract concept. For example, in a Development Studies (DS) classroom, problem-solving techniques should be embedded within topics such as population and development.

Since problem-solving skills are mental processes that cannot be directly observed, it is essential to identify observable measures to assess individual proficiency. Adeoye et al. (2023) indicate that learners with advanced problem-solving skills typically perform better on questions requiring such abilities. Bloom (1969) proposed that verbs such as "appraise," "synthesise," "analyse," "infer," "assess," and "suggest" are indicators of higher-order thinking and problem-solving abilities. Learners proficient in these skills demonstrate deep understanding, seek clarification, ask pertinent questions, and request necessary assistance (Bloom, 1969).

Ahuja (2021) identifies self-directed learning as a characteristic of learners with well-developed problem-solving skills. Marshall (2022) views self-directed learning as an approach that fosters creativity, autonomy, and independent decision-making. The Ministry of Education and Training (MoET, 2018) emphasises independent learning in Development Studies, linking it to problem-solving. Thus, self-directed learning is integral to teaching and learning in this context.

Bloom's taxonomy of verbs serves as an indicator of problem-solving skills. Furthermore, Bloom (1956) developed a taxonomy outlining six cognitive levels, ranging from simple to complex thinking. Armstrong (2010) and Berger (2020) observe that educators often focus on lower-order thinking questions, which do not effectively foster higher-order thinking skills. In the same way, Bloom (1969) argued that the initial levels of "remembering" and "understanding" are foundational for progressing to higher levels, such as "creating" and "analysing." Table 1 below illustrates Bloom's taxonomy of verbs.

BLOOM'S LEVEL	DEFINITION	KEY VERBS
Create	Creating new patterns	Formulate, develop, create,
		evaluate, invent,

Table 2.1 Bloom's Taxonomy of Verbs

Evaluate	Making judgments on the	Evaluate, assess, describe,
	value of evidence	determine, support, argue,
		examine
Analyse	Breaking down information	Analyse, classify, associate,
	into small parts and	criticise, categorise
	identifying any relationships	
Apply	Application of relevant	Calculate, solve,
	procedures or ideas to a	demonstrate, determine,
	situation	illustrate
Understand	Understanding the meaning of	Explain, paraphrase, give,
	facts	summarise, discuss
Remember	Recognising, retrieving and	List, recite, outline, define,
	recalling facts	name, match, quote, label
Bloom (1969)		

The table presented above delineates the levels in Bloom's taxonomy, their meanings, and the associated key verbs. Clark (2015) emphasised that a lack of variation in teaching and learning methods adversely affects learners' comprehension of concepts and the development of essential skills. This concern motivated Bloom to create a taxonomy of verbs designed to enable educators to diversify lesson objectives and assessment strategies, thereby fostering the development of higher-order thinking skills in learners (Armstrong, 2010; Clark, 2015).

Bloom's taxonomy, as outlined by Bloom (1969), transitions from low-order thinking to higherorder thinking through the levels of applying, analysing, evaluating, and creating. This framework assists teachers in planning lessons that promote advanced cognitive processes among students. It is evident that the effective development of problem-solving skills necessitates a meticulously planned lesson aligned with the taxonomy of verbs. Such alignment is crucial for achieving favourable outcomes in the cultivation of higher-order thinking skills.

2.5 Impact of traditionalised learning on learner's acquisition of skills

Scholars have reported that traditional teaching and learning methods have been prevalent since ancient times. Nonetheless, some prominent philosophers, such as Paulo Freire (1970) and John

Dewey (1986), view these methods as rigid and inflexible (Hu, 2024). According to Serin (2018), traditional teaching and learning describe a classroom scenario where the teacher exerts full control while learners passively receive information. Moreover, Hu (2024) illuminate that traditional teaching, often referred to as teacher-centered learning, focuses primarily on knowledge transmission to ensure that learners master their subject matter. Essentially, such a learning environment is criticised for depriving learners of opportunities to engage in creative thinking and to connect their learning experiences with real-world situations.

In his "banking approach," Freire (1970) argued that traditional instruction is disconnected from real-life contexts. He likened this educational style to a banking system, where teachers deposit information into learners, who are perceived merely as containers for that knowledge (Wang, 2015). Freire (1970) strongly opposed this method of instruction. Both Connell (2012) and Wang (2015) supported Freire's critique, indicating that teacher-centered learning hinders the development of essential skills such as problem-solving and critical thinking. They pointed out that in such environments, learners are typically discouraged from asking questions or engaging in peer interactions; instead, they passively absorb information and follow the teacher's directives.

Conversely, Dewey (1986), as a proponent of progressive education, advocates for learnercentered teaching over teacher-centered methods. Dewey contends that passive reception of knowledge limits learners' problem-solving abilities (Wang, 2015). Progressive educators, therefore, argue that allowing learners to explore and participate in hands-on activities enhances their learning experiences (Singh, 2023). In contrast, traditional teaching methods are criticised for impeding learners' understanding of concepts and the development of critical skills, as they treat learners as blank slates. Consequently, the adoption of alternative learning methodologies in the classroom can support learners in developing crucial abilities such as problem-solving.

2.6 Empirical review

A study that is well-established connects with previous research through agreement or disagreement regarding findings, methodologies, and theories. Notably, the present research incorporates empirical literature to provide evidence on the use and implementation of Inquiry-Based Learning (IBL). According to Eke (2022), an empirical review involves analysing earlier studies to offer a comprehensive understanding of a particular concept. Consequently, this section

discusses reviewed studies that utilised guided inquiry-based learning as a teaching strategy and examines its implementation.

Nisa et al. (2018) investigated the efficacy of guided inquiry-based learning in enhancing critical thinking skills among 90 senior high school learners in Indonesia. The findings revealed that, at the pre-test stage, learners demonstrated poor performance in physics static fluid. However, following the intervention with guided inquiry-based learning, learners' performance improved progressively across all three groups involved. Nisa et al. (2018) supported these results with Vygotsky's (1978) assertion that learners perform better when actively engaged. This active involvement, facilitated by guided inquiry-based learning, also appeared to enhance information retention in long-term memory, thereby fostering critical thinking, problem-solving, and mastery of concepts. Thus, it can be concluded that guided inquiry-based learning supports critical thinking in subjects such as physics and is applicable in other areas to improve higher-order thinking skills.

Similarly, Orosz et al. (2023) conducted a mixed-method study in Hungary with grade 9 chemistry learners, which demonstrated that guided inquiry-based learning significantly impacts learners' problem-solving skills. The results indicated poor performance at the pre-test level, while post-test results showed that guided inquiry, through continuous scaffolding, positively supported problem-solving skills among learners following the chemistry curriculum. This suggests that when implemented effectively, guided inquiry-based learning can enhance the teaching and learning of subjects such as Development Studies (DS). Orosz et al. (2023) recommended that teachers should elucidate ideas and offer guidance, especially when employing new teaching methods like inquiry. These findings align with those of Nisa et al. (2018).

In the context of improving problem-solving skills, Sukontawaree et al. (2022) explored the effectiveness of IBL and cooperative learning in grade 4 within the Thai educational setting. Their study found that IBL positively impacted learners' problem-solving abilities, with an improvement from 14.28% to 54.14%, thus benefiting 80% of learners in solving problems. The positive impact was attributed to IBL's ability to address barriers such as deficits in group work skills and lack of knowledge in using scientific instruments. Importantly, the teacher's role as a facilitator during the inquiry process highlighted that IBL is a learner-centered approach that fosters problem-solving skills.

Regarding teachers' perspectives on IBL, Simbarashe and Gherda (2019) conducted a study in Zimbabwe, reporting that teachers lacked the knowledge and competencies required to implement IBL effectively in advanced-level physics classrooms. Teachers expressed concerns about the demands of public examinations, limited human resources, and inadequate time for effective teaching. As a result, they recommended ongoing training for teachers on implementing IBL and support from relevant stakeholders to align public examination requirements with IBL practices. These results are consistent with Mandina's (2024) study, which explored teachers' perceptions and implementation of IBL in rural Zimbabwean schools, indicating that proper guidance and support are essential for successful IBL implementation.

Significantly, Mandina (2024) demonstrates that while chemistry teachers in Zimbabwe had a positive attitude towards IBL, its implementation faced challenges due to factors such as overcrowded classrooms, poorly resourced schools, and insufficient skills for preparing and implementing IBL. These findings highlight that resource constraints are a significant barrier to effective IBL implementation.

In Tanzania, Mwenda and Ndayambaje (2021) conducted a quasi-experimental study to evaluate the impact of inquiry-based teaching (IBT) on biology students' academic performance. The results in this study indicated that the IBT-treated group outperformed the expository group, with mean scores of 56.21 compared to 49.31, respectively. Characteristics of IBL, such as cooperative learning, self-directed learning, and active involvement, were linked to enhanced student performance in IBT. This aligns with previous research on IBL, which suggests that IBL is an effective teaching strategy that yields favourable outcomes when applied in the classroom.

Similarly, Masilo (2018) examined the use of IBL to improve grade 11 learners' problem-solving abilities in Gauteng, South Africa. Her research utilised a mixed-method design and revealed that IBL significantly improved students' problem-solving skills, with the experimental group demonstrating a substantial effect size of 0.55 compared to the control group. Masilo (2018) attributed these improvements to IBL's capacity to facilitate active engagement, cooperation, interaction, and self-directed learning. Pedaste et al. (2015) further supported these findings, noting that IBL helps learners achieve a deep understanding of concepts and problems, thereby enhancing their problem-solving abilities.

In Lesotho, Koekoe (2023) studied geography teachers' perceptions of IBL, finding that while teachers had a positive attitude towards IBL, its implementation faced challenges such as the reliance on summative assessments that do not align with IBL and the need to complete the syllabus within a specified timeframe. Teachers also reported a lack of training as a barrier to effective IBL implementation. These results are consistent with the findings of Gherda (2019) and Shadreck (2024), which identified similar constraints in Zimbabwe.

Conversely, Gholam (2019) found that students at the American University in Dubai did not hold negative views about IBL, nor did they experience issues such as a lack of engagement or frustration with inquiry activities. Instead, Gholam (2019) identified systemic factors within the school as significant obstacles to IBL implementation. Therefore, it is evident that the challenges faced in implementing IBL are not unique to Lesotho but are also present in other educational contexts.

2.7 Gap and Appraisal

In the literature examined, researchers investigated and endorsed IBL as an appropriate technique for teaching and acquiring high-order abilities such as problem-solving and critical thinking. Furthermore, the majority of IBL research has been conducted in other nations, with an emphasis on science disciplines. In Africa, including Lesotho, studies on IBL have primarily focused on perception rather than implementation. Although various studies have been undertaken, there is little research on the use of IBL in teaching DS or social studies, and Lesotho is no exception. Thus, the current study aims to fill the knowledge and pedagogic gap on the effective use of IBL in teaching and learning of DS in Lesotho.

2.8 Conceptual framework

A conceptual framework is defined by Luft et al. (2022) as an overview that researchers present to demonstrate their understanding of the study's primary concepts while also establishing linkages between the variables under consideration. Including this section in the current study was deemed advantageous to clarify the relationships between the variables in question. In research, a variable is defined as a component or foundation that the researcher can modify, observe, or control based on their research interests (Abiodun-Oyebanji, 2017; Nwankwo & Emunemu, 2014).

While various types of variables exist in research, this study specifically focused on dependent, independent, and moderating variables. According to Abiodun-Oyebanji (2017) and Bhandari (2023), a dependent variable is an outcome variable that reflects the supposed effect or consequence of an independent variable. Furthermore, Bhandari (2023) suggests that an independent variable, also referred to as an input variable, is manipulated by the researcher to influence the dependent variable. A moderating variable, as defined by Nwankwo and Emunemu (2014), is one that enhances, weakens, or cancels out the relationship between two or more variables.

In the current study, guided IBL was chosen as the key independent variable, as it directly impacts the dependent variable. Learners' problem-solving skills were considered the dependent variable because they were instrumental in determining the effect of guided IBL. Other concepts evaluated as independent variables in this study included the IDEAL model, collaboration, active learning, and mediating tools. Although these variables were not of primary interest, they helped to establish the correlations between the main independent variable (guided IBL) and the dependent variable (problem-solving abilities). Prior knowledge was deemed a moderating variable due to its potential to either strengthen or diminish the relationship between the variables under consideration. The framework in Figure 2.3 illustrates the relationships between the variables in this study.


The framework in Figure 2.3 illustrates the relationship between the dependent and independent variables in this study. The IDEAL model, which served as the theoretical foundation, is closely linked to guided inquiry-based learning (IBL), as it reinforces and provides guidelines for integrating IBL phases in the classroom. Bransford and Stein (1984) developed the IDEAL model with the intent of offering step-by-step guidance to help teachers effectively teach problem-solving skills to students. Thereupon, this model facilitated the adoption of guided IBL in the classroom due to the simplicity of its acronym, IDEAL, which is notably easier for educators to remember.

Furthermore, the framework highlights the connection between guided IBL, the primary independent variable, and collaboration. Guided IBL, rooted in constructivism, is inherently learner-centered. This instructional approach enables learners to engage with and exchange ideas while embracing their diverse perspectives (Pedaste et al., 2015). Prominent educational theorists such as Freire (1970) and Dewey (1966) have advocated for learner collaboration in the classroom, emphasising that it is the teacher's responsibility to ensure learners have opportunities to exchange ideas during the learning process.

Vygotsky (1978) also underscored the importance of collaboration as a form of scaffolding, where more knowledgeable learners assist their peers in functioning within their zone of proximal development. This underscores the high correlation between guided inquiry-based learning and collaboration in enhancing learners' problem-solving abilities. Gates (2018) further supports this by viewing collaboration as a crucial factor in developing higher-order thinking skills, including critical thinking and problem-solving. For instance, when learning about urbanization in a Development Studies (DS) class, students were encouraged to interact and pose questions. As they engaged in discussions, those with a better understanding of urbanisation helped others who struggled with the concept.

Active engagement in the classroom is essential for learners to acquire specific skills. Dewey (1966) asserted that knowledge is best acquired through action, emphasising that students who actively participate in the learning process tend to comprehend more than those who do not. The close relationship between collaborative learning, guided IBL, and active learning is evident in Figure 2.3. To facilitate learners' ability to ask questions and work collaboratively, they must be

viewed as active producers of knowledge. Freire (1970) argued that students who are given the opportunity to participate actively in their education develop critical skills and deepen their conceptual understanding. This suggests that collaboration and active learning are significantly interconnected in this study, impacting learners' problem-solving skills.

Mediating tools also played a significant role as independent variables in this study. For example, learners were shown YouTube videos and graphics to help them explore ideas and formulate questions. These mediating tools were crucial in promoting collaboration and active participation among learners. When asked to identify the causes of pollution on natural resources, learners used their textbooks and YouTube videos to gather information, which aided them in providing accurate answers. Saeidi and Ahmadi (2016) argue that learning materials are effective in engaging learners with the visual information presented to them.

While the aforementioned variables appear to have a strong relationship in influencing changes in the dependent variable—learners' problem-solving skills—the moderating variable of prior knowledge was also considered impactful. Specifically, learners with some prior knowledge of a particular topic found it easier to collaborate and actively participate in the learning process. Although some learners lacked prior knowledge, their peers shared information with them during collaborative activities. As a result, these learners became more active and engaged, asking questions throughout the inquiry process. It is therefore evident that prior knowledge played a role in either strengthening or weakening the relationship between the independent variables and the dependent variable (learners' problem-solving skills).

In conclusion, this section discussed the relationship between the primary independent variable (guided IBL) and the dependent variable (learners' problem-solving abilities). It also demonstrated how other contributing independent variables, such as collaboration, relate to the primary independent variable to enhance its influence on the dependent variable. Finally, the moderating variable of prior knowledge was shown to be crucial in either strengthening or weakening the correlation between the study's variables.

2.9 Chapter Summary

This chapter provided a comprehensive review of the literature relevant to the current investigation. Additionally, previous studies related to this research were examined to identify gaps in the existing literature. The relationships between the variables in this study were also established in this chapter. Notably, this section played a crucial role by offering a theoretical foundation that informed the implementation of guided inquiry-based learning (IBL) in the Development Studies (DS) classroom.

CHAPTER 3 3.0 Methodology 3.1 Introduction

This section presents the methodology employed in the current investigation. According to Cohen et al. (2018), methodology in academic research is a crucial component, serving as a roadmap for how the entire study will be conducted. In this study, the methodology was particularly relevant as it provided a structured approach to addressing the research questions. Resultantly, the following presentation outlines the steps undertaken in this investigation, beginning with an exploration of the research paradigm.

3.2 Paradigm

A philosophical underpinning in academic research is critical as it clarifies the epistemological, ontological, and methodological stance of the study (Creswell & Plano Clark, 2011). According to Lincoln et al. (2011), a paradigm comprises assumptions that define the researcher's worldview, representing a set of beliefs that guides the research process. Jansen (2023) identifies three key paradigms: interpretivism, positivism, and pragmatism. This study was guided by the philosophical lens of positivism.

The positivist paradigm aims to establish causal links between the study's variables (Park et al., 2019). Specifically, it facilitates the determination of the relationship between the dependent variable (the outcome) and the independent variable (the intervention). As outlined in the previous chapter, a dependent variable in research is defined as an outcome that reflects changes due to the independent variable (Bhandari, 2023; Helmestine, 2024). Conversely, the independent variable is the factor that the researcher manipulates to observe its effect on the dependent variable (Bhandari, 2023). In this study, guided inquiry-based learning (IBL) was the independent variable, with the aim of assessing its impact on the dependent variable—learners' problem-solving skills.

The positivist paradigm was instrumental in identifying the relationships between the intervention (guided IBL) and the outcome (learners' problem-solving skills). Alharahsheh and Pius (2020) emphasise that the positivist paradigm is grounded in the belief that empirical evidence is paramount and that knowledge should be derived from neutral, measurable, and quantifiable

observations. This paradigm seeks to generate numerical data that can be analysed statistically to produce generalisable and replicable results (Jansen, 2023). Guided by this paradigm, the study recorded learners' performance on problem-solving tasks to gather evidence on the impact of guided IBL.

Furthermore, positivism was particularly suitable for this study because the primary objective was to collect data that is free from human bias—data that is observable, measurable, and quantifiable, thereby ensuring replicability. Positivists maintain an ontological stance of realism, implying that a single, observable reality exists independent of human bias (Cohen et al., 2018). Epistemologically, positivists believe that knowledge is objective, observable, and quantifiable (Park et al., 2019). Whereof, the methodology of this paradigm involves using structured designs and instruments to collect and analyse data statistically, thereby reaching valid conclusions (Jansen, 2023).

Accordingly, the positivist paradigm provided a framework for gathering numerical data to address the research questions and draw conclusions about the impact of guided IBL on teaching problemsolving in DS. Creswell and Plano Clark (2018) argue that replicable, observable, and quantifiable results lend rigor to research. Moreover, Park et al. (2019) suggest that when the researcher's role is confined to data collection and analysis, without influencing the responses, the data remains unbiased, leading to robust results. The replicability of research findings is crucial, as it enhances the validity and reliability of the study (Cohen et al., 2018). Thus, the positivist paradigm was essential, not only as a blueprint for the research process but also for ensuring the production of reliable and valid results. The next section presents the research approach.

3.3 Research approach

A research approach refers to the methods and strategies employed in a study to facilitate data collection, aligning with the nature and objectives of the investigation (Cohen et al., 2018). Creswell and Creswell (2018) identified three primary research approaches: qualitative, quantitative, and mixed methods. For the current study, the quantitative approach was selected.

In accordance with, Bhandari (2023), the quantitative approach involves the use of structured instruments to collect numerical data, which is subsequently analysed using statistical methods.

This approach is particularly suited to studies that aim to establish causal relationships between variables, resulting in observable and quantifiable outcomes (Cohen et al., 2018). In the context of this study, the quantitative approach was essential for utilising statistical tools to gather numeric, quantifiable, and observable data concerning the impact of guided inquiry-based learning (IBL) on learners' problem-solving skills in DS. For that reason, statistical instruments such as the Likert scale, observation checklists, and tests were employed to ensure the rigour and validity of the findings.

The next section discusses the research design utilised in this study.





The research design employed in this study, as illustrated in the figures referenced earlier, serves as a critical framework for guiding the overall approach of the investigation. Conforming to McCombes (2021), a research design acts as a blueprint that directs the entire study process. Building on this, Creswell and Plano Clark (2018) identify various types of quantitative research designs, including survey, non-experimental, quasi-experimental, descriptive, and experimental designs. In light of these options, the present study adopted a quasi-experimental design featuring two groups—a pre-test and a post-test.

Importantly, Cohen et al. (2018) describe quasi-experimental designs as experimental investigations where randomisation is challenging due to pragmatic or ethical constraints. Given these challenges, the quasi-experimental design was deemed the most suitable approach for this study. Specifically, in Lesotho, where not all schools offer Development Studies (DS) as a subject,

the difficulty in randomly assigning participants became apparent. Therefore, a single high school in Maseru that offers this subject was deliberately chosen for the study.

Furthermore, Thomas (2024) highlights the advantages of this design, particularly its high level of external validity, which is attributed to the use of real-world interventions rather than controlled laboratory experiments. Thus, in this study, guided inquiry-based learning (IBL) was implemented as a practical intervention. To facilitate a comprehensive comparison, the study considered two groups—the experimental and control groups—as depicted in Figures 3.1 and 3.2. This comparison aimed to assess the effects of guided IBL on enhancing learners' problem-solving abilities relative to traditional teaching methods, thereby underscoring the relevance and applicability of the chosen research design.

3.5 Population, Sample, and Sampling Techniques

3.5.1 Population

In academic research, defining the population is crucial as it forms the basis for drawing generalisable conclusions. According to Bhandari (2023) and Cohen et al. (2018), the population refers to the entire group of individuals that the researcher aims to study and make inferences about. Additionally, Creswell (2018) underscores that the importance of identifying the population lies in its capacity to enable the generalisation of findings from a selected sample to the broader group. In this study, the population consisted of three Grade 11 classes from a high school in Maseru. Specifically, two of these classes comprised 20 learners each, while the third class contained 23 learners. This delineation of the population ensures that the study's findings can be applicable to a larger group of Grade 11 students.

3.5.2 Sample

While the population encompasses the total group from which conclusions are drawn, the sample represents a subset of this population from which data is collected (McCombes, 2023). Thus, sampling is pivotal in research as it facilitates the collection of data from a manageable group that can be generalised to the entire population (Bhandari, 2023). For this study, a sample of two intact Grade 11 classes from the Maseru high school was selected to represent the overall Grade 11 population concerning the impact of guided Inquiry-Based Learning (IBL) on enhancing problem-

solving skills in Development Studies (DS). Specifically, the sample comprised 40 students, with the experimental group including 20 learners (7 males and 13 females) and the control group also consisting of 20 learners (9 males and 11 females). This sampling strategy ensures a representative analysis of the impact of guided IBL.

3.5.3 Sampling Techniques

Sampling techniques play a critical role in determining how participants are selected for a study. According to McCombes (2023), these techniques can be categorised into probability sampling and non-probability sampling. Given that this study employed a quasi-experimental design where randomisation was not feasible, non-probability sampling was deemed the most appropriate method. Cohen et al. (2018) specify that non-probability sampling involves the selection of participants in a non-random manner. In support of this, Bhandari (2023) explains that non-probability sampling selects participants based on convenience, preference, or the researcher's judgment. Therefore, purposive sampling, utilising intact classes, was employed in this study.

Purposive sampling involves selecting participants based on specific characteristics or judgments made by the researcher (Creswell & Creswell, 2018). Furthermore, intact sampling, a subtype of purposive sampling, refers to using pre-existing groups without altering their composition (Ali, 2024; Swami, 2016). These techniques were found to be suitable for this study due to the fact that not all schools in Lesotho offer DS as a subject. As a result, the school that offered DS was purposively selected. Moreover, the Grade 11 intact classes were chosen because they were expected to have a better understanding of DS concepts, which would facilitate their engagement in the guided inquiry process. Importantly, these classes were used in their entirety, without being divided or altered (intact classes), thereby ensuring the study's integrity.

3.6 The Choice of the School and Its Profile

This section outlines the characteristics of the school where the intervention was conducted. The selected school, situated in the Maseru district of the southern region, offers Development Studies as one of its subjects. The school accommodates approximately 800 learners and 25 teachers and provides a range of subjects, including Fashion and Design, Agriculture, Development Studies, Accounting, Sesotho, and English. The Grade 11 learners were purposively included in the study

because they were believed to possess a more advanced understanding of DS concepts, which would enhance their ability to engage in the inquiry-based learning process.

Upon arriving at the school, the principal reported that the institution previously had a strong performance record in DS, although this had recently declined. This observation presented an opportunity to implement guided IBL and assess its potential impact on learners' problem-solving skills.

3.7 Data Collection Instruments

This section delineates the instruments utilised for data collection in the study, providing a comprehensive overview of their roles and applications.

3.7.1 Pre-test

To address the first research question, which aimed to assess the current level of problem-solving skills among learners in Development Studies (DS), a pre-test was administered prior to the intervention. As noted by Budert-Waltz et al. (2023), a pre-test functions as a diagnostic tool designed to evaluate learners' prior knowledge or current status before the application of the treatment. Both the control and experimental groups participated in this pre-test. This initial assessment was crucial in establishing a baseline for comparing the effects of guided Inquiry-Based Learning (IBL) with traditional teaching methods in enhancing problem-solving skills.

3.7.2 Observation Checklist

In response to the second research question, which investigated the impact of guided IBL on learners' problem-solving skills in DS, an observation checklist and a post-test were employed. The observation checklist was specifically utilised for the experimental group that engaged in the guided IBL treatment. Cohen et al. (2018) define an observation checklist as a structured tool used to evaluate specific behaviours or attributes. In this study, the checklist was designed to assess learners' behaviours during the guided IBL intervention. The checklist included statements aligned with the five phases of IBL, and each statement was rated as either "yes" or "no" to reflect the extent to which the treatment influenced learners' behaviour. A Development Studies teacher from

the study site observed the intervention sessions in the experimental group, while the researcher facilitated the process.

3.7.3 Post-test

Subsequent to the treatment, a post-test was administered to evaluate whether guided IBL had affected learners' problem-solving skills. This assessment served as a follow-up to the behaviours recorded on the observation checklist. According to Stratton (2019), a post-test is conducted after participants have received a specific treatment to measure its impact on the dependent variable— in this case, learners' performance in problem-solving tasks. Both the experimental and control groups completed the post-test, and the results were compared with pre-test outcomes to determine the effectiveness of guided IBL relative to traditional teaching methods.

3.7.4 Likert Scale

To address the third research question, which sought to measure learners' satisfaction with guided IBL, a five-point Likert scale was utilised. This scale, applied exclusively to the experimental group, included ratings from "strongly disagree" to "strongly agree." Cohen et al. (2018) describe the Likert scale as a tool for assessing various factors such as satisfaction, attitudes, and opinions, using a range of statements rated on a scale. The Likert scale used in this study incorporated both positive and negative statements to gauge learners' satisfaction with the guided IBL approach. The inclusion of varied statements was intended to ensure consistent and reliable responses from learners as they selected the options that best reflected their views.

The following section will detail the procedures implemented to carry out the inquiry process within the classroom setting.

3.8 Treatment procedure

The process detailed in this section outlines the implementation of guided Inquiry-Based Learning (IBL) for teaching Development Studies (DS) students how to solve problems. Specifically, the experimental group was the sole cohort subjected to the guided IBL procedure, while the control group received instruction through the conventional teaching method. The guided IBL treatment was conducted over a four-week period, with five lessons scheduled each week. Notably, two of

these lessons were allocated 80 minutes each, and the remaining three lessons were allotted 40 minutes each. This lesson allocation was consistent for both the experimental and control groups.

During the intervention, the topic of population and development was taught, following the advice of the co-teachers in accordance with their scheme of work. Prior to the commencement of the intervention, both the control and experimental groups underwent a pre-test. Additionally, before the guided IBL treatment began, learners in the experimental group were oriented on the principles of guided IBL. This orientation included guidance on expected behaviours, roles, responsibilities, and the teacher's role during the inquiry process.

Throughout the guided IBL intervention, lessons in the experimental group were observed by another DS teacher from the participating school, while the researcher facilitated the inquiry process. In contrast, the control group was instructed using the conventional method. In this traditional approach, learners were required to listen to the teacher, take notes as dictated, and complete tasks individually. No classroom observations were conducted for the control group.

In the implementation of guided IBL with the experimental group, the five illustrative phases of IBL, as delineated by Pedaste et al. (2015), were utilised. These phases were discussed in detail in the previous chapter. Figure 3.3 on the following page provides an illustration of these phases, followed by a detailed description of how the inquiry process was executed.



Figure 3.3 The steps engaged in teaching problem-solving using guided IBL

STEP 1

The orientation phase represents the initial step in the Inquiry-Based Learning (IBL) process. During this phase, learners were encouraged to voice any questions or thoughts they had regarding the subject matter. This stage was crucial for assessing learners' prior understanding of the day's topic. By drawing on the learners' existing knowledge, the orientation phase helped guide the direction of the inquiry process and clarify the topic being discussed. A key component of this phase involved prompting learners to articulate the day's topic in their own words, thereby facilitating their comprehension. For instance, when introducing the concept of overpopulation, learners posed questions such as, "What happens when there are too many people in one area?" "What factors lead to overcrowding?" and "How do resources get affected when there are more people than usual in one area?"

STEP 2

Following the orientation phase is the conceptualisation phase. In this stage, the development of the inquiry question or problem of investigation occurs. Building upon the questions generated by learners during the introduction of the topic in phase one, learners, with continued support from the teacher, formulated the inquiry question. This stage also involved setting specific goals and outlining the procedures necessary to address the inquiry question. For example, in the lesson where overpopulation was introduced, the inquiry question focused on the problems caused by overpopulation and potential strategies for mitigating these issues. The problem of investigation was derived from the questions posed in the first phase. Additionally, learners devised a procedural outline to describe the problems caused by overpopulation and strategies to address them. This procedure was displayed in bullet points on the board and included definitions, contributing factors, and the impact on natural resources.

STEP 3

The third step in the IBL process is the investigation phase, where evidence is gathered. In this phase, learners utilised the procedures developed in the previous stage to collect relevant information related to the inquiry problem. For instance, in the lesson on overpopulation, learners gathered data from textbooks (LGCSE Development Studies) and online sources. During this

phase, learners were encouraged to work collaboratively in groups to share and discuss ideas pertinent to the topic and the investigation problem. My role as the teacher was pivotal in presenting visual aids, such as pictures related to population and overpopulation, to facilitate a clearer understanding of the subject matter (see Figure 3.4).



Figure 3. 4 Example of the pictures that were presented to learners

https://www.coggnserve-energy-future.com/causes-effects-solutions-of-overpopulation.pghpg

As learners collected information, they were encouraged to ask questions and collaborate both within their groups and with other peer groups. This collaborative approach facilitated the gathering of more comprehensive information and enabled learners to find answers to their questions. The teacher provided guidance on how to organise the information in relation to the problem of investigation. Based on the earlier phases, learners defined overpopulation as "the situation whereby the total number of people living in a particular area exceeds its carrying capacity." Additionally, they identified factors contributing to overpopulation as "high fertility rates," "migration," "tourism," and "lack of access to modern contraceptives."

STEP 4

The demonstration phase follows the collection of information. In this phase, learners attempted to provide solutions to the problem of investigation based on the information gathered. For instance, in the context of overpopulation, learners suggested solutions such as implementing social campaigns on resource management to mitigate the misuse of natural resources. They also proposed that empowering women and educating both men and women about the importance of contraceptives could help reduce overpopulation.

This phase also served to demonstrate whether the inquiry process effectively assisted learners in formulating solutions to the problems. According to Pedaste et al. (2015), it is crucial for the teacher to continually support and encourage learners to understand the concept thoroughly before attempting to solve the problem. This support plays a significant role in helping learners develop problem-solving skills as they gain a comprehensive understanding of the issues related to the topic. As exemplified, learners first defined overpopulation and related issues before proposing solutions.

STEP 5

The final phase is the discussion, which involves assessing the skills acquired through the inquiry process. In this phase, learners were given individual assessments, including a short exercise, to evaluate whether they had developed problem-solving skills through the guided inquiry process. This assessment was accompanied by guidance and ongoing support from the teacher. Additionally, learners reflected on the procedures undertaken, including question-asking and information gathering, to identify areas for improvement in future inquiries.

In conclusion, the IDEAL model, represented by its acronym, was instrumental in implementing guided IBL. It provided a structured framework that facilitated learners' recall of each phase of the IBL process. Moreover, during the intervention, observations were made on learners' behaviors while they engaged with the guided IBL. Following the intervention, all learners from both the experimental and control groups were administered a post-test to evaluate the effectiveness of the guided IBL approach.

3.9 Data analysis methods

Data analysis encompasses various techniques utilised to identify patterns, trends, and insights within a dataset (Taherdoost, 2020). This process is crucial in research as it provides a comprehensive overview of the statistical tools or methods employed to draw conclusions from

the data (Chia, 2023). In this study, I employed both descriptive and inferential statistics using the Statistical Package for the Social Sciences (SPSS, Version 26) software. SPSS is a robust statistical tool developed by IBM, designed to facilitate complex statistical analyses, including regression, ANCOVA, MANOVA, t-tests, and correlations. Its user-friendly interface allows researchers to organise and present data precisely (Williams, 2024).

Inferential statistics, as described by Stapor (2020), are employed to make predictions and generalisations about a larger population based on sample data. In contrast, descriptive statistics are used to summarise and describe the characteristics of the dataset, including measures of central tendency (mean, median, and mode), variability, and distribution (ibid).

In this study, statistical tests in SPSS were conducted at a confidence level of 95% and a significance level of 0.05. A 95% confidence level is commonly used in research as it provides a standard benchmark for decision-making regarding statistical results (Beavans, 2023). The significance level of 0.05 is particularly important as it indicates whether the results are statistically significant. Typically, this significance level is associated with a 95% confidence interval (Hayes, 2024).

The significance level is often represented by a p-value, which denotes the "probability value." A p-value less than 0.05 (<0.05) is considered significant, indicating that observed differences in the results are likely to be meaningful (Beavans, 2023; Menon, 2024; Virag, 2024). Conversely, a p-value greater than 0.05 (>0.05) suggests that there are no significant differences between the observed groups' results (Menon, 2024). In essence, the p-value serves as a threshold for determining the statistical significance of the observed findings.

3.9.1 Pre-test data analysis

To analyse the data obtained from the pre-test on learners' current level of problem-solving skills in Development Studies (DS), an independent sample t-test was conducted to calculate the mean and significance levels of the groups prior to treatment. This analysis was crucial as it enabled the identification of statistical differences between the two groups, providing insights into their problem-solving skills before the intervention. Establishing these differences was essential for assessing the homogeneity of the groups. According to Kulkarni and Patil (2022), ensuring that the groups being compared are homogeneous, or possess equal variance, is fundamental. Homogeneity implies that the groups have similar characteristics before the treatment is applied. This uniformity is vital in minimising the effects of confounding variables that could otherwise skew the results.

Furthermore, identifying group differences before the treatment allows the researcher to attribute any observed changes to the specific intervention applied. As Kulkarni and Patil (2022) assert, this approach helps in clearly determining whether the intervention had an effect. Resultantly, the independent sample t-test was instrumental in assessing the differences between the control and experimental groups before the treatment commenced, thereby providing a baseline for evaluating the impact of the intervention.

3.9.2 Observation Checklist Analysis

To analyse the data obtained from the checklist, which recorded information on the influence of guided Inquiry-Based Learning (IBL) on learners' problem-solving skills, descriptive statistics were employed. Specifically, frequencies were calculated to determine the percentages of responses rated as "yes" and "no" concerning the statements related to observed behaviour. The calculation of frequencies is crucial in research, as it facilitates the presentation of data in a clear and comprehensible manner, often through frequency tables or charts (Kaplan, 2024; Rhodes, 2023). This approach enabled a detailed assessment of how guided IBL affected learners' problem-solving skills in Development Studies (DS), whether positively or negatively.

3.9.3 Post-Test Analysis

The data collected from the post-test, which complemented the observation checklist, were analysed using Analysis of Covariance (ANCOVA). ANCOVA was employed to compare the performance of learners from both the experimental and control groups regarding their problem-solving skills in DS. According to Shieh (2020), ANCOVA is a statistical method used to determine differences between group means while accounting for potential confounding variables. This technique enhances the traditional Analysis of Variance (ANOVA) by addressing the limitations of ANOVA, such as its failure to account for outliers.

Consequently, ANCOVA was deemed appropriate for analysing the post-test results as it incorporates the influence of covariates. A covariate is defined as a factor that varies with the response variable and potentially impacts it (Ford, 2023). In this analysis, the pre-test scores were treated as a covariate to control for their potential effect on the post-test outcomes, which measure learners' problem-solving skills following the intervention. This method was chosen for its ability to compute means and p-values effectively, thereby evaluating the extent to which guided IBL influenced learners' problem-solving skills in DS. For a visual representation, refer to Figure 3.5



Figure 3. 5 An illustration of how the analysis was carried out

Figure 3.5 above illustrates the execution of the Analysis of Covariance (ANCOVA) in SPSS, aimed at assessing the impact of guided Inquiry-Based Learning (IBL) on learners' problem-solving skills in Development Studies (DS). As depicted, the pre-test scores were designated as a covariate to account for their influence on the relationship between the intervention and learners' problem-solving skills post-treatment. This approach was crucial for controlling any potential effects that the pre-test might have on the outcomes.

In this analysis, the post-test scores were treated as the dependent variable, as they reflect the effect of the guided IBL on learners' problem-solving skills. By analysing the post-test scores, the study was able to determine the impact of the treatment, measured through test scores, on the learners' skills in DS. This methodological setup ensured that the assessment of the intervention's effectiveness was accurate and reliable.

3.9.4 Likert scale analysis

To address the final research question, frequencies were calculated to gauge learners' satisfaction with the use of guided Inquiry-Based Learning (IBL) in the classroom. This approach was chosen because frequency counts provide a clear and immediately understandable summary of the data. The results are displayed in a frequency table, which presents learners' responses, as measured by a five-point Likert scale, for each phase of IBL.

The subsequent section will discuss the assumptions that were verified prior to conducting the analysis.

3.10 Assumptions that were followed before running the SPSS tests

It is crucial to adhere to specific assumptions before conducting parametric tests in SPSS. Parametric tests, by definition, rely on the assumption that the population data is normally distributed to ensure the effectiveness and accuracy of the results obtained (Campbell & Shantikumar, 2016; Campbell & Swinscow, 2009). In the current study, the parametric tests employed were the independent sample t-test and the Analysis of Covariance (ANCOVA).

To ensure the validity of these tests, several assumptions were verified beforehand. Firstly, it was confirmed that both the dependent variable and the covariate were measured on a continuous scale, as illustrated in Figure 3.6 below.

ta p	re-test po	ost test analy	/sis D	ESSERTATIO	N.sa	v [DataSet1]	1 - IBM SPSS S	Statistics	Data Editor						
<u>F</u> ile	<u>E</u> dit	View D	ata	Transform	n	<u>A</u> nalyze	<u>G</u> raphs <u>U</u>	tilities	Extensions	<u>W</u> indow	<u>H</u> elp				
		Name		Туре		Width	Decimals		Label	Values	Missing	Columns	Align	Measure	Role
	1	pretest		Numeric		8	0	Pre-te	st	None	None	8	를 Center	🛷 Scale	🦒 Input
	2	posttest		Numeric		8	0	Post-t	est	None	None	8	를 Center	🛷 Scale	🦒 Input
	3	group		Numeric		8	2	Group		{1.00, Exper	None	8	■ Right	\delta Nominal	🦒 Input
	4														
	5														
	6														
	7														

Figure 3. 6 Variable view of dataset loaded in SPSS

The above figure demonstrates that the dependent variable, when conducting the t-test and ANCOVA, was measured on a continuous scale. This is evident from the column labelled "measure," where the post-test was recorded as a scale. Additionally, the first assumption for ANCOVA stipulates that the covariate must also be measured on a continuous scale (Field, 2018).

As shown in the same column, the pre-test, which served as the covariate, was also treated as a scale, thus meeting this assumption prior to running the tests.

The second assumption for both ANCOVA and the independent sample t-test is that the independent variable should consist of two or more categorical groups (Field, 2018). In this study, the independent variable was represented by the group, determined by the type of treatment or intervention administered. Hence, this assumption was satisfied as both the control and experimental groups were involved, with each group receiving the appropriate intervention.

Furthermore, the parametric tests require homogeneity of variance among the groups involved (Campbell & Shantikumar, 2016). To verify this, Levene's test for equality of error variances was conducted. The results indicated that equal variance was assumed across the groups, as the p-value obtained was greater than 0.05, specifically 0.895, confirming that the groups were homogenous.

Finally, Field (2018) notes that the dependent variable should be approximately normally distributed for each group of the independent variable, ensuring accurate results. To meet this assumption, the Shapiro-Wilk test was employed. Values below 0.05 indicate an abnormal distribution, while values greater than 0.05 suggest normal distribution. An illustration of this test result is provided below.

	Kolm		Shapiro-Wilk			
Group	Statistic	Df	Sig.	Statistic	Df	Sig.
Experimental	.188	20	.063	.903	20	.046
Control	.170	20	.133	.968	20	.707

Table 3.1 The normality test

Table 3.1 presents the Shapiro-Wilk test results, showing significance values of 0.46 for the experimental group and 0.707 for the control group, both of which exceed the threshold of 0.05. This suggests that the dependent variable was approximately normally distributed within each group. In consquence, the accuracy of the results obtained from running the ANCOVA in this study is affirmed.

Additionally, the Kolmogorov-Smirnov test further supports this conclusion, as it also yielded significance values greater than 0.05 for both groups. Specifically, the significance value for the experimental group was 0.063, while for the control group it was 0.968.

This section is crucial as it confirms that the tests used in this study were conducted appropriately. Adhering to these assumptions ensures the validity and robustness of the results. Therefore, it can be concluded that the data in this study was normally distributed, reinforcing the reliability of the findings.

3.11 Validity and Reliability

Reliability is defined by Middleton (2023) as an indicator of how consistent and replicable a chosen method is. In contrast, validity refers to the extent to which an instrument measures what it is intended to measure (Carroll & Goodfriend, 2023). To ensure reliability, the statistical tests used to analyse data with SPSS software were conducted at a 95% confidence level, a measure that enhances the replicability and consistency of the results (Mark, 2020). This confidence level supports the reliability and robustness of the findings in this study. Furthermore, parametric assumptions, including tests of normality, were conducted prior to the analysis using SPSS software. These assumptions are crucial for ensuring reliability, as they indicate the population distribution, which contributes to producing true and consistent results.

For internal validity, the study employed ANCOVA as a data analysis method. Field (2012) reports that ANCOVA effectively controls for extraneous variables, thereby enhancing internal validity. External validity was addressed by using guided inquiry-based learning (IBL) to improve learners' problem-solving skills, rather than relying on a laboratory-based treatment that may not be applicable to real-world situations. According to Bhandari (2023), real-world interventions like guided IBL add rigor to a study because they can be more easily generalised to other contexts. Content validity was ensured by aligning the pre-test and post-test instruments with the DS grade 11 syllabus to confirm their relevance and accuracy. Additionally, these tests were reviewed by other DS teachers to verify whether they effectively measured learners' problem-solving skills. Content validity pertains to how well an instrument represents the construct it is designed to measure (Bhandari, 2023).

3.12 Ethical Considerations

Ethics is crucial for enhancing the validity of research. According to Patton (2015), ethical considerations involve adhering to guiding principles to ensure valid and robust findings. In this study, ethical standards such as informed consent were meticulously observed. Participants and their parents were informed about the study's purpose and benefits. Consent forms were distributed to parents to ensure they were aware of their children's participation in the treatment and observation. Additionally, participants were assured of anonymity and confidentiality regarding their personal details, and they were informed that their involvement in the study was voluntary. To gain access to the school, an introductory letter from the Faculty of Education, under the Department of Language and Social Education (LASED), was presented to the principal to explain the research interest. Finally, to maintain academic integrity, the Turnitin plagiarism checker was utilised to confirm that the work was free from plagiarism.

3.14 Chapter Summary

This chapter outlined the methodology employed in the study, which utilised a quantitative quasiexperimental design involving both control and experimental groups to teach problem-solving in DS. It detailed the instruments used for data collection and the measures implemented to ensure rigorous results. In summary, this chapter served as a comprehensive blueprint for the entire study, detailing the methodological framework and procedures that underpin the research.

Chapter 4

4.0 Results Presentation and Interpretation

4.1 Introduction

The previous chapter addressed the methodological aspects of the study. This chapter focuses on presenting and analysing the data collected from the pre-test, observation checklist, post-test, and Likert scale. As outlined earlier, the analysis was conducted using SPSS version 26 software. It is important to reiterate that the independent sample t-test and ANCOVA were performed at a 95% confidence level and a significance level of 0.05, as specified in the previous chapter. Hayes (2024) notes that a 95% confidence level in statistical analysis indicates the degree of certainty with which we can infer that the results derived from the sample reflect the broader population in the study.

Descriptive statistics were computed using SPSS, with the results represented through frequency percentages. Frequencies were chosen for their clarity and ease of interpretation. This chapter begins with an overview of the demographic profile of the participants, which provides insight into the types of individuals involved in the study and their characteristics.

The study aimed to address the following research questions:

- 1. What is the current level of problem-solving skills among learners in DS?
- 2. What influence does guided inquiry-based learning (IBL) have on learners' problemsolving skills in DS?
- 3. What is the level of satisfaction among learners regarding the use of guided IBL in DS classes?

In addressing these questions, the results are organised around several key themes: the current level of problem-solving skills among learners in DS, the impact of guided IBL on learners' problem-solving abilities, post-test results, and learner satisfaction with guided IBL in DS classes. The following section presents the demographic characteristics of the participants involved in the study.

4.2 Demographic Information of Participants

This section outlines the characteristics of the participants in the study. Such demographic information is crucial for ensuring ethical research practices and for understanding the participant group (Bartschi, 2020). According to Bartschi (2020), providing detailed demographic profiles is an ethical obligation, as it helps researchers consider diversity and inclusivity, thereby minimising biases and enhancing the robustness of the conclusions. Additionally, demographic information provides context for the generalisability of the results. Tables 4.1 and 4.2 present the demographic characteristics of the participants in this study.

Age		Gender	
group	Male	Female	Total
15-17	4	6	10
18-20	2	4	6
21-22	1	3	4
TOTAL	7	13	20

 Table 4.1 Profile for Experimental group

Age	Gender

 Table 4.2 Profile for the Control group

Age		Gender	
group	Male	Female	Total
15-17	5	7	12
18-20	2	2	4
21-22	2	2	4
TOTAL	9	11	20

Table 4.1 presents the characteristics of learners in the experimental group. Of the 20 learners, 7 were boys and 13 were girls. The age distribution of these learners is also detailed. Specifically, among the 7 male learners, four were aged between 15 and 17 years, and two were aged between 18 and 20 years. Additionally, one male learner was aged between 21 and 22 years.

In contrast, among the 13 female learners in the experimental group, six were aged between 15 and 17 years, four were aged between 18 and 20 years, and three were aged between 21 and 22 years.

Table 4.2 provides the demographic characteristics of the control group. This group also comprised 20 learners, with 9 males and 11 females. Among the male learners, five were aged between 15 and 17 years, two were aged between 18 and 20 years, and two were aged between 21 and 22 years. On the female side, seven learners were aged between 15 and 17 years, two were aged between 18 and 20 years, and two were aged between 21 and 22 years.

In summary, the study involved a total sample of 40 learners, comprising 24 females and 16 males.

4.3 The current level of problem-solving among learners in DS

To address Research Question One, an independent sample t-test, along with frequency calculations, was employed to assess the existing level of problem-solving skills among learners in DS prior to the intervention. The independent sample t-test was particularly instrumental in identifying differences between the control group (which utilized the traditional method) and the experimental group (which engaged in guided inquiry-based learning, or IBL) before the treatment commenced. This comparative approach was essential for ensuring that the results accurately reflected the impact of guided IBL on problem-solving skills in DS.

The learners' mark scores were utilised to calculate both the mean scores and the significance levels within each group. The application of statistical tools, such as SPSS software, in quantitative research helps ensure the validity of results by minimising human bias during calculations (Mark, 2020).

Table 4.3 presents the frequencies and percentages of learners' pre-test scores. Initially, the data reflects the overall performance of learners on problem-solving questions without distinguishing between the groups (refer to Table 4.3).

Mark scores	Number of learners (N=40)	Percentage (%)
1	6	15%
2	5	12.5%
3	6	15%
4	2	5%
5	8	20%
6	4	10%

Table 4.3 The mark scores of learners at the pre-test lev

7	4	10%
8	5	12.5%

Table 4.3 provides a detailed breakdown of the pre-test scores for learners, which were evaluated out of a maximum of 20 points. The first column lists the specific test scores achieved by the learners, while the second column indicates the number of learners who received each score. The third column presents the percentage of learners who achieved each corresponding score.

From the data, it is evident that, out of the 40 learners who completed the pre-test, none achieved a score of 10 or higher, which was set as the passing mark. Specifically, 15% of learners scored only 1 mark out of 20. Additionally, 12.5% of learners achieved a score of 2 marks. Notably, 15% of learners scored 3 marks on the test.

Continuing with the results, 5% of learners attained a score of 4 marks, 20% scored 5 marks, and 10% scored 6 marks. Furthermore, another 10% of learners received 7 marks. The highest mark obtained was 8, achieved by 12.5% of learners. These results indicate that overall, the problem-solving skills of the learners in DS are notably low, as none reached the 50% threshold considered a passing mark. The scores ranged from as low as 1 mark to a maximum of 8 marks out of 20, reflecting a generally poor level of problem-solving skills among the learners.

To provide a more comprehensive analysis of their problem-solving abilities, the mean score and standard deviation (SD) were calculated, and these results are detailed in Table 4.4.

	Table 4. 4 7	The mean score of	of learners in	DS who	took a	pre-test
--	--------------	-------------------	----------------	--------	--------	----------

	Ν	Mean		Std. Deviation	
	Statistic	Statistic	Std. Error	Statistic	
Pre-test	40	4.35	.371	2.348	
Valid N (listwise)	40				

Table 4.4 shows the mean score and standard deviation (SD) for a sample of 40 learners prior to the treatment. The mean score was m=4.35, and the SD was SD=2.348. These figures reveal a notably low level of problem-solving skills among the learners in DS.

Despite these results, it remains challenging to ascertain whether there were any differences in problem-solving skills between the control and experimental groups based solely on the pre-test scores. As a result, to determine if there were any significant differences between the groups before the intervention, an independent sample t-test was conducted. This statistical test was employed to assess potential disparities in problem-solving abilities between the learners in the control and experimental groups prior to the treatment.

Table 4.5 An independent sample t-test of the control and experimental groups beforetreatment

		Levene's Equa	s Test for llity of							
		Varia	ances			t-te:	st for Equalit	y of Means		
									95% Co	nfidence
									Interva	l of the
						Sig. (2-	Mean	Std. Error	Differ	ence
		F	Sig.	Т	Df	tailed)	Difference	Difference	Lower	Upper
Pre-	Equal variances	.149	.702	133	38	.895	100	.752	-1.623	1.423
test	assumed									
	Equal variances			133	37.976	.895	100	.752	-1.623	1.423
	Equal variances			133	37.976	.895	100	.752	-1.623	1

The results presented in Table 4.5 indicate that there were no significant differences in problemsolving skills between the control and experimental groups at the pre-test stage. Specifically, Levene's test of equality of variances confirmed that equal variances could be assumed between the two groups. According to Mark (2020), a p-value greater than 0.05 in an independent sample t-test suggests that the assumption of equal variance is valid. Thus, the results in the first column of the table should be interpreted.

Variance, as defined by Thompson and Wesolowski (2018), refers to the numerical distribution used to assess differences between groups or compare distributions. In this study, the p-value

(denoted as "sig.") in Table 4.5 was greater than 0.05, supporting the conclusion that there were no significant differences between the control and experimental groups at the pre-test. Specifically, the p-value was p=0.702, indicating that the variance between the groups was similar.

Furthermore, Mark (2020) emphasised that when the 95% confidence interval for the difference in means includes both negative and positive values, it indicates no significant difference between the groups' performances. In this study, the confidence interval ranged from -1.623 to 1.423, reinforcing the conclusion that the prior problem-solving skills of learners in DS were equivalent across the groups at the pre-test.

The analysis further revealed that the problem-solving skills of learners, as measured by their test scores, were notably poor in both the experimental and control groups, as reflected in Table 4.4. The independent sample t-test was instrumental in demonstrating that both groups had comparable levels of problem-solving skills before the implementation of guided IBL. Whereof, it was concluded that the control and experimental groups were homogeneous prior to the treatment. For further clarification, Table 4.6 provides the mean scores of each group at the pre-test stage.

				Std.
			Std.	Error
Group	Ν	Mean	Deviation	Mean
Experimental	20	4.30	2.408	0.539
Control	20	4.40	2.349	0.525

Control

 Table 4.6 The mean scores of Control and Experimental groups before treatment

As illustrated in Table 4.6, the mean score for the experimental group prior to treatment was m=4.30, while the mean score for the control group was m=4.40. This minor difference of 0.10 indicates that the two groups were statistically comparable at the pre-test stage. As a result, it can be inferred that the level of problem-solving skills among learners in DS was similar in both groups before the treatment was applied.

These findings suggest that learners in both groups exhibited poor problem-solving skills, particularly in tasks requiring such skills. This observation is consistent with the data presented in Table 4.3, where no learner achieved a score of 50%, equivalent to 10 out of 20. Such results corroborate the earlier assertion that the two groups were homogeneous at the pre-test, given that equal variance was assumed between them. Thus, the analysis confirms that both groups entered the treatment phase with similar baseline problem-solving abilities.

4.4 The influence of guided inquiry-based Learning on learners' problem-solving skills in DS

The second research question aimed to investigate the influence of guided inquiry-based learning on the problem-solving skills of learners in DS. To address this, an observation checklist was utilised during the intervention, and a post-test was administered following the intervention. Specifically, the observation checklist was employed throughout the teaching process using guided inquiry-based learning. It is important to note that classroom observations were conducted exclusively with the experimental group—the focus of this study—where changes in learners' problem-solving skills were anticipated.

Observations were conducted during the first lesson, the third lesson, and the final day of the intervention. The selection of these specific instances was intended to monitor changes in learners' behaviour and their problem-solving skills as they engaged with the inquiry process. Prior to beginning the observations, I provided a comprehensive explanation of the inquiry process and outlined the expected behaviours and phases of inquiry-based learning to the learners.

The observation checklist comprised ten statements, with each phase of inquiry being represented by four statements. The observer recorded responses by ticking "yes" or "no" based on their observations. The resulting data from these checklists are summarised in frequency tables reflecting the observer's judgments. Table 4.7 presents the analysis of the first observation checklist, detailing the initial findings.

Phases of inquiry-	Statements (4	Percentage	Statements	Percentage
based learning	per phase)	(%)	(4 per	(%)
	(yes)		phase)	
			(n o)	
Orientation	1	25%	3	75%
Conceptualisation	2	50%	2	50%
Investigation	3	75%	1	25%
Conclusion	0	0%	4	100
Discussion	3	75%	1	25%

Table 4.7 An observation checklist for the first lesson on guided inquiry-based learning

Table 4.7 presents the observation data across the five phases of inquiry, detailing the number of statements and their respective percentages. In the orientation phase, 75% of the statements were marked as "no," while only 25% were marked as "yes." This suggests that, during the orientation phase, learners exhibited some interest in the lesson, but their familiarity with the inquiry process was limited, as it was a new concept for them. For instance, a statement assessing learners' ability to identify the problem of investigation when posing questions was marked as "no," indicating that learners struggled to understand their tasks at this stage.

In the conceptualisation phase, the results were more balanced, with 50% of the statements marked as "no" and the remaining 50% marked as "yes." This phase is crucial as it involves setting goals and procedures to address the problems identified in the previous phase. The fact that 50% of the statements were marked as "yes" indicates that learners exhibited some inquiry features, such as collaboration with peers. This suggests that, although learners were starting to adapt to the inquiry process, challenges persisted. For example, a statement regarding learners' ability to identify relevant materials was marked as "no," reflecting difficulties in locating resources needed for information retrieval during the initial lesson.

The investigation phase, as shown in Table 4.7, revealed that 75% of the statements were marked as "yes," while 25% were marked as "no." This phase involves exploring solutions to the identified problems. The high percentage of "yes" responses indicates that learners were more engaged in

collaborative activities, such as sharing ideas with peers. However, they faced challenges in seeking relevant information to solve the problems effectively, as reflected in the "no" responses. An example from this phase includes an activity where learners worked in groups to assess possible solutions for reducing global warming in Lesotho. Despite their efforts, difficulties in gathering pertinent information were evident.

Overall, these observations highlight that while learners showed some progress in collaboration and engagement with the inquiry process, significant challenges remained in effectively identifying and utilizing resources to address the problems presented. The accompanying figure provides a visual representation of the information learners generated during their group activities in the investigation phase.

o reduce global warming the government in Lesothon must put policies whereby people will know that wrong, you ng more trees when

Figure 4.1 An illustration of learners' responses on a given activity

The figure above demonstrates that learners exhibited a limited understanding of environmental issues and struggled to apply practical solutions to the problems presented. Instead of exploring information that could directly address the activity, learners proposed general actions without clearly articulating how the information gathered could be utilised to tackle global warming. This suggests that, during the initial lesson, learners encountered difficulties in developing effective solutions for the problem at hand.

In the fourth phase, the conclusion phase, Table 4.7 reveals that learners were unable to provide solutions to the identified problems, as indicated by the fact that none of the four statements were marked as "yes." Specifically, the percentage for statements marked as "no" was 100%, reflecting a complete lack of success in this phase.

Conversely, in the final phase, the discussion phase, 75% of the statements were marked as "yes," indicating that learners were able to reflect on their shortcomings during the inquiry process. However, 25% of the statements were marked as "no," suggesting that learners partially struggled to complete this stage effectively. Notably, observations showed that learners failed to deliver a detailed summary of the solutions proposed for the identified problems, primarily due to their earlier difficulties in identifying the problem of investigation.

Overall, the results presented in Table 4.7 indicate that learners faced significant challenges during the initial inquiry lesson, as it was their first experience with a problem-solving approach through guided inquiry. My role was crucial in providing substantial guidance and support to help learners engage effectively in the inquiry process, including identifying problems from the given text or lesson content and developing solutions.

The subsequent presentation of the observation checklist results for the third week of the intervention will offer further insights into the learners' progress and continued challenges.

Phases of inquiry-	Statements	Percentage (%)	Statements	Percentage (%)
based learning	(4 per phase)		(4 per phase)	
	(yes)		(no)	
Orientation	3	75%	1	25%
Conceptualisation	4	100%	0	0%
Investigation	3	75%	1	25%
Conclusion	4	100%	0	0%
Discussion	4	100%	0	0%

Table 4.8 Results from observation checklist of the third lesson of the intervention

Table 4.8 presents the results of the observation checklist for the third week of the intervention. A comparison of these results with those from the initial observation reveals a marked improvement. Specifically, during the orientation phase, 75% of the checklist statements were marked as "*yes*," whereas only 25% of the statements, which equates to one statement, were marked as "*no*." This indicates a significant change in learners' behaviour as they became more familiar with the inquiry process. According to the checklist, the only challenge encountered during the orientation phase was learners' difficulty in relating their questions to real-world scenarios.

In the second phase, conceptualisation, learners demonstrated notable success in executing the inquiry process by establishing goals and procedures to address the identified problem. As shown in Table 4.8, all statements in this phase were marked as *"yes."* This suggests that, with ongoing support and guidance from the teacher, learners became adept at formulating appropriate procedures for problem-solving and understanding the necessary steps to follow.

Regarding the investigation phase, 75% of the statements were marked as *"yes,"* while 25% were marked as *"no."* These results represent a substantial improvement from previous observations, as learners displayed a greater understanding of guided inquiry and were more effective in exploring solutions to the identified problems. Moreover, at this stage, learners exhibited curiosity by posing questions that facilitated the search for suitable solutions to the investigation problem.

For example, when tasked with describing solutions to address the infant mortality rate, learners asked pertinent questions such as *"What happens if a baby is born underweight?"* and *"What measures can ensure that pregnant mothers consume a balanced diet?"*

In the conclusion phase, all four statements were affirmatively marked as "yes," signifying that learners successfully applied the information acquired in preceding phases to resolve the problems presented. This outcome, as illustrated in Table 4.8, reflects a 100% success rate for this phase. Such a high percentage indicates a progressive enhancement in learners' problem-solving skills through guided inquiry, as they grew increasingly adept at navigating the nature of their lessons.

Furthermore, Table 4.8 reveals that learners exhibited substantial improvement during the discussion phase. Each statement evaluating learners' behaviour in demonstrating the development of problem-solving skills was marked as *"yes,"* resulting in a 100% success rate for this phase.

This observation suggests that by the third week, learners had markedly improved their capacity to engage in reflective discussions and articulate solutions effectively.

In summary, Table 4.8 shows that throughout the third week of the intervention, most statements across all phases were marked affirmatively as "yes." Moreover, the proportion of statements marked as "no" ranged from 0% to 25%, indicating that only one out of four statements was marked as "no." These results underscore a significant enhancement in learners' problem-solving skills within the inquiry-based classroom compared to previous assessments.

The observer's note on the checklist remarked that *"learners demonstrate the ability to solve problems. Their ability to ask questions shows that they are accustomed to the process of inquiry."* This observation suggests that increased familiarity with the inquiry process has positively influenced learners' classroom behaviour, contributing to improved problem-solving capabilities.

Finally, the observation checklist for the last lesson of the intervention was examined to assess learners' behaviour on their final day of participating in guided inquiry. This conclusive assessment was pivotal for evaluating the impact of inquiry-based learning on learners' problem-solving skills. Figure 4.1 on the subsequent page presents the results of this observation



Figure 4.2 An Observation Checklist for the last lesson of the intervention

Figure 4.2, presented above, displays the results from the observation checklist utilised during the final lesson of the intervention. The data reveal that learners exhibited exemplary behaviour across all phases of inquiry on this day. Specifically, in the orientation phase—the initial stage—all four statements were marked as *"yes,"* as previously indicated. This observation suggests that learners, at this phase, effectively demonstrated the capacity to formulate insightful questions aimed at identifying the problem under investigation.

Furthermore, the results for the conceptualization phase show that none of the statements were marked as "no"; rather, all four statements were affirmatively marked as "yes." This pattern is consistent across the investigation, conclusion, and discussion phases as well. Consequently, these findings underscore that the observed improvements in learners' behaviour can be attributed to the implementation of guided inquiry-based instruction. The results imply that such an approach not only facilitates learners' engagement with the inquiry process but also enhances their ability to reflect on and improve their inquiry skills for subsequent lessons.

The previously presented observations had indicated some challenges faced by learners during their inquiry process. However, Figure 4.2 demonstrates that, by the final lesson, learners had successfully navigated the inquiry process, suggesting they had reflected on and addressed their earlier difficulties, potentially with the assistance of their teacher. This improvement is indicative of the effectiveness of guided inquiry-based learning in fostering learners' development and adaptation. An example of the exercise administered to learners during the final class of the intervention is illustrated below

Solutions to problems caused by overpoints In Lesotho Jervie Juch as hospitals and foot gards (an be narely to the page/ due age Sautoous Clinics lan be built at tural areas so that Ropk from rural greas do not more to Mrban areas and lause everyopulations for - fail rarres lace he prised to that who do not have feed bud those the are not employed to reduce hegh better services Nemen and girls can be polleted about the importance of Some amissions which causes air populion Ean be reduced by charging every one the use of Confraceptues to that they will not have an amount when they emost smokes, for example the factories/ (An be charged for many ballies emillion of smoke, That will make them to emit less smoke which coulded pollytion

Figure 4. 3 Learners' exercise on problem-solving

The figure above illustrates the responses of learners to an exercise administered in a classroom setting on the topic of overpopulation. Learners were tasked with evaluating potential solutions to the challenges posed by overpopulation. Their responses revealed a certain level of understanding and demonstrated their capacity to propose viable solutions to the identified problems. This exercise was instrumental in confirming that as learners become more accustomed to the process of inquiry, their performance and behaviour in problem-solving progressively improve.

In summary, the three observation checklists presented indicate a notable improvement from the first lesson of the intervention to the final one. Specifically, during the initial lesson, learners encountered difficulties in solving problems within a guided inquiry-based learning framework. However, as the intervention progressed, they increasingly succeeded in providing solutions to the problems presented. This improvement is evidenced by the observer's comment regarding the learners' behaviour in the inquiry-based classroom.

The observer noted that "learners can ask questions and provide appropriate solutions, which in turn facilitated collaboration and led to more effective responses to the questions posed." This observation suggests that with proper guidance and sustained support, learners were able to engage successfully in the inquiry process and generate solutions to the issues related to the topic at hand.

4.5 Post-test results presentation

Following the intervention described above, a post-test was administered to evaluate the impact of guided inquiry-based learning on learners' problem-solving skills in DS. This post-test was conducted for both the control and experimental groups, with the objective of comparing the outcomes to assess the influence of

traditional versus inquiry-based learning approaches on these skills. To analyse the data, the Analysis of Covariance (ANCOVA) was performed using SPSS version 26 software. The ANCOVA facilitated the computation of key statistical measures, including the mean, standard deviation, and p-values. The results of this analysis are presented in Tables 4.9 and 4.10.

Group	Mean	Std. Deviation	Ν
Experimental	13.35	3.689	20
Control	6.15	3.031	20

 Table 4.9 The mean scores of learners after treatment

Table 4.9 presents the mean scores and standard deviations reflecting the performance of learners following the administration of the treatment to each group. As shown, the experimental group achieved a mean score of M = 13.35, SD = 3.689, whereas the control group had a mean score of M = 6.15, SD = 3.031. These results indicate that the experimental group significantly outperformed the control group on the post-test. This finding contrasts with the results in Table 4.6, where minimal or no difference was observed between the mean scores of learners in both the experimental and control groups during the pre-test.

However, it is important to note that the calculation of the mean does not account for confounding variables that may influence the accuracy of the results obtained. Resultantly, SPSS provides both the mean table and the ANCOVA summary table to address this issue. Below is the ANCOVA summary of learners' post-test performance, which determines whether the treatment had a significant effect on learners' problem-solving skills after controlling for the confounding variable. In this analysis, the pre-test, as discussed in the previous chapter, was treated as a covariate, as it was expected to co-vary with the outcome variable, represented by the post-test scores
	Type III Sum of					
Source	Squares	Df	Mean Square	F	Sig.	Partial Eta Squared
Corrected Model	553.362 ^a	2	276.681	25.713	.000	.582
Intercept	565.724	1	565.724	52.574	.000	.587
Pretest	34.962	1	34.962	3.249	.080	.081
Group	523.979	1	523.979	48.695	.000	.568
Error	398.138	37	10.760			
Total	4754.000	40				
Corrected Total	951.500	39				

Table 4.10 The ANCOVA summary of the test scores of experimental and control groups

In Table 4.10, the most significant values, which provide insight into the impact of the treatment on the dependent variable, are highlighted in light blue. However, it is also important to explain the other factors represented in the table. Firstly, the corrected model shows the results after adjusting for the effect of covariates on the dependent variable (Howell, 2017). This adjustment helps to clarify the true effect of the independent variable, guided inquiry-based learning, on the dependent variable, which in this case is learners' problem-solving skills. The intercept represents the expected value of the dependent variable when the covariates are set to zero (Howell,2017). Additionally, the pre-test reflects whether there was a significant difference between the groups before the treatment was administered (Pallant, 2016).

The "group" variable represents the actual impact of the independent variable on the dependent variable (Field & Miles, 2012). Specifically, the value under "group" pertains to both the control and experimental groups. The results here are instrumental in understanding the potential of guided inquiry-based learning (IBL) to enhance problem-solving skills among learners in DS. The type III sum of squares indicates the amount of variance explained by each variable while accounting for other variables (Howell, 2017; Pallant, 2016). The degrees of freedom (D.F.) represent the number of independent pieces of information available. Meanwhile, the mean square provides an estimate of variance within groups (ibid). The F-value is the ratio that compares the variability within groups (Field & Miles, 2012), and the significance value (Sig.) determines whether the

results are statistically significant relative to the chosen alpha value, also known as the p-value (Beavans, 2024). Finally, partial eta squared represents the strength or the effect size of the independent variable on the dependent variable (Pallant, 2016).

The results in Table 4.10 indicate that guided inquiry-based learning had a positive impact on learners' problem-solving skills in DS. The level of significance for the group value in the table was p = 0.00, which is less than 0.05, thereby showing that the results are significant. This finding suggests that after learners were exposed to different teaching methods—guided IBL and traditional methods—their performance differed significantly, with the significant improvement attributable to guided inquiry-based learning. This conclusion is supported by the value of the pretest in Table 4.11, where the recorded significance was p = 0.80, which is greater than 0.05. Thus, the pre-test results indicate no substantial difference between the groups before the intervention. Further, as demonstrated in Table 4.10, the mean score for the experimental group was higher than that of the control group, confirming the effectiveness of the intervention.

Statistically, the impact of the treatment on learners' problem-solving skills is presented as follows: F(1,37) = 48.695, p < 0.05. These results suggest that at the post-test level, the scores for the intervention and comparison groups differed significantly, confirming that guided IBL had a positive impact on enhancing learners' problem-solving skills. The significance value of p < 0.05 (0.00 < 0.05) further illustrates that when learners engage in inquiry-based processes, their problem-solving abilities improve more than when they engage in traditional learning methods. The strength of the association between guided inquiry and learners' problem-solving skills, as indicated by the partial eta squared (n = 0.568), suggests a moderate effect size, which is highlighted in blue in the row representing the group statistics.

According to Brydges (2019), effect sizes can be categorised as small, medium, or large, with corresponding ranges of approximately 0.2, 0.5, and 0.8. It is important to emphasise that the ANCOVA results in Table 4.11 helped to ensure the accuracy of these findings by controlling for the effect of the covariate (pre-test), which had the potential to distort the association between the treatment and the dependent variable (problem-solving skills, as measured by post-test scores). Therefore, it can be concluded that guided inquiry-based learning has significant potential to enhance learners' problem-solving skills in DS compared to conventional teaching methods.

4.6 The level of satisfaction among learners in DS on the use of guided inquiry-based learning in teaching problem-solving skills.

The third research question focused on assessing learners' satisfaction with the use of guided inquiry-based learning in DS. To capture their responses, a five-point Likert scale was employed, consisting of 10 statements from which learners could indicate their level of satisfaction. It should be noted that this Likert scale was administered exclusively to the treatment group, comprising 20 learners. Table 4.11 presents the frequency percentages corresponding to learners' responses. For the purpose of calculating the median, responses ranging between 1 and 2.5 were categorized as "disagree," while those falling between 2.6 and 5 were categorized as "agree."

The analysis of median values was conducted using SPSS software (version 26). It is noteworthy that the level of satisfaction percentages was calculated based on responses from the 20 learners in the treatment group, as illustrated below. The calculation of the median was deemed highly appropriate for analysing learners' responses. As discussed in the previous chapter, the median calculation is particularly valuable because it has the potential to control for outliers that could otherwise affect the validity of the results. Accordingly, it was considered essential to validate the frequency percentages of learners' responses through the calculation of the mean, ensuring that the results presented are robust and reliable

<u>Number of learners who responded</u> = Level of satisfaction (%)

Total number of learners involved

Above is the formular that was used to calculate the percentages in Table 4.11.

The values that indicate the degree of agreement were five thus the scale was a five-point Likert scale. The Likert Scale points were abbreviated as follows: SD=Strongly Disagree, D=Disagree, N=Neutral, A=Agree and SA=Strongly Agree.

Statement		Level of	Median				
		SD	D	N	А	SA	value
1.	Guided IBL helped me to grasp	0%	0%	15%	60%	25%	4
	difficult concepts more easily						
2.	With guided inquiry, I was able to	0%	10%	15%	30%	25%	4
	apply solutions to real-world						
	problems						
3.	I found it interesting to be given an	0%	0%	10%	25%	65%	4
	opportunity to ask questions during						
	the process of inquiry						
4.	I felt overwhelmed and confused	40%	40%	15%	5%	0%	2
	during the process of inquiry						
5.	Guided inquiry encouraged	0%	5%	5%	30%	30%	5
	collaboration among us in the						
	classroom						
6.	Collaboration with my classmates	30%	30%	10%	20%	10%	2
	challenged me as others had differing						
	opinions						
7.	I enjoyed being actively involved in	0%	0%	10%	50%	40%	4
	the classroom as I had a better						
	understanding of the topic taught						
8.	In guided inquiry class, I was unable	35%	30%	15%	10%	10%	2
	to choose best solutions to the						
	identified problems						
9.	I like being continually supported in	0%	0%	15%	50%	35%	4
	the process of learning						
10.	I was able to identify solutions to the	0%	0%	20%	55%	25%	4
	problems independently in guided						
	IBL classroom						

Table 4.11 Learners' satisfaction on the use of guided inquiry-based learning

As presented in Table 4.11, learners were generally satisfied with the use of guided inquiry-based learning (IBL) in their classroom for developing problem-solving skills. Notably, 60% of learners agreed with the first statement on the scale, indicating that guided IBL helped them grasp difficult

concepts in the DS classroom. Additionally, 25% of learners strongly agreed with this statement. Meanwhile, 15% of learners neither agreed nor disagreed, remaining neutral. The table further shows that none of the learners strongly disagreed with the statement, implying that all respondents, except those who remained neutral, marked their responses as "agree" or "strongly agree," indicating a majority consensus on the positive impact of guided IBL.

The data in Table 4.11 reveal a median of four for the first statement, which was categorised as "agree" regarding learners' satisfaction with guided IBL. This finding reinforces the notion that most learners perceived guided IBL as effective in enhancing their problem-solving skills. Moreover, 0% of learners strongly disagreed with the second statement, and 10% of learners disagreed. On the contrary, 15% of learners (three out of 20) remained neutral, while 30% agreed that guided inquiry helped them apply solutions to real-world problems. Similarly, 25% of learners strongly agreed that guided inquiry played an important role in allowing them to provide solutions to real-world problems. This suggests that a substantial number of learners—approximately 15 out of 20—expressed satisfaction with the use of guided inquiry in their classroom. The median value for the second statement also indicated general agreement, suggesting that learners were satisfied with guided inquiry for learning problem-solving skills in DS.

Regarding the statement "I found it interesting to be given an opportunity to ask questions," learners generally agreed that they were satisfied. Specifically, 65% of learners (13 out of 20) strongly agreed that they found it interesting to be given an opportunity to ask questions in an inquiry-based class as they sought solutions to problems. Additionally, 25% of learners agreed with this statement, indicating their satisfaction with the inquiry process that allowed them to ask questions. Notably, 10% of learners remained neutral on this statement, while 0% disagreed or strongly disagreed, suggesting widespread satisfaction with the opportunity to ask questions in a guided IBL classroom. The median of four further supports this conclusion, statistically indicating significant agreement on learners' satisfaction.

In response to the fourth statement, a large proportion of learners disagreed that guided IBL overwhelmed or confused them during problem-solving. Specifically, 40% of learners strongly disagreed, and another 40% disagreed, totalling 80% of learners who disagreed that guided inquiry was confusing or overwhelming. Again, 15% of learners remained neutral, and only one learner

agreed that guided inquiry was confusing and overwhelming, with 0% strongly agreeing. The median of two, categorised as "disagree," suggests that learners generally did not find guided inquiry to be confusing or overwhelming.

The fifth statement on the Likert scale, "Guided inquiry encouraged collaboration among us in the classroom, which helped us provide solutions to problems more easily," received positive responses. As presented in Table 4.11, 0% of learners strongly disagreed, while only one learner (5%) disagreed with the statement. Importantly, 30% of learners agreed that collaborating with peers helped them develop problem-solving skills, while 60% strongly agreed. The median of five, indicating high satisfaction, suggests that learners found collaboration beneficial for problem-solving in a guided IBL classroom.

When asked about their satisfaction with the statement, "Collaboration with my classmates challenged me as others had differing opinions which confused me," learners generally disagreed. As indicated in Table 4.11, 30% of learners strongly disagreed, and another 30% disagreed with the statement. However, 20% of learners agreed that collaboration posed a challenge due to differing opinions, and 10% strongly agreed, while 10% remained neutral. Despite some learners acknowledging challenges, the median of two, categorized as "disagree," suggests that learners were generally satisfied with collaboration in guided IBL.

Regarding the seventh statement, "I enjoyed being actively involved in the classroom as I had a better understanding of the topics taught," the results indicate high satisfaction among learners. The median of four, categorised as "agree," suggests a strong level of satisfaction. None of the learners strongly disagreed or disagreed with the statement, while 10% remained neutral, 50% agreed, and 40% strongly agreed, indicating that active involvement in the classroom through guided IBL contributed to their understanding and satisfaction.

In response to the eighth statement, "In a guided inquiry class, I was unable to choose the best solutions to the identified problems," learners generally disagreed. Specifically, 35% of learners strongly disagreed, and 30% disagreed, indicating that they did not perceive guided IBL as inhibiting their ability to choose the best solutions. Only 15% of learners remained neutral, while 10% agreed and another 10% strongly agreed. The median of two, indicating disagreement,

suggests that learners were satisfied with their ability to choose solutions in a guided IBL classroom.

For the ninth statement, "I like being continually supported in the process of learning," the results show strong satisfaction among learners. None of the learners strongly disagreed or disagreed with the statement, although some (15%) remained neutral. A significant majority—50% of learners— agreed, and 35% strongly agreed that they liked receiving continuous support during the learning process. The median of four indicates that learners generally agreed, suggesting that ongoing support during guided IBL contributed to their satisfaction.

Finally, in response to the last statement, the results indicate that learners were satisfied with the use of guided IBL in their classroom. Specifically, 25% of learners strongly agreed that they were able to identify solutions to problems independently in a guided inquiry-based classroom, while 55% agreed. Only 20% of learners marked their responses as neutral, and 0% disagreed or strongly disagreed. The median of four indicates that learners were generally satisfied with the use of guided IBL, particularly in fostering independent problem-solving skills

4.7 Chapter summary

The results presented in this chapter indicate that learners' problem-solving skills were initially at the same level before the intervention, as reflected in the pre-test scores. Principally, none of the learners involved in the study scored at least 50%, equivalent to 10 out of 20. However, following the administration of a specific treatment to each group, those taught using guided inquiry-based learning (IBL) demonstrated superior performance compared to those instructed through conventional methods. Observations further revealed that learners' behaviour and performance in problem-solving activities progressively improved as they became more accustomed to the guided inquiry process.

Additionally, the findings suggest that a significant number of learners were satisfied with the use of guided inquiry in their classroom. Although some learners chose not to express their views on certain statements, the number of those who agreed with the positive statements exceeded those who remained neutral. Only a small minority of learners found collaboration during the inquiry process to be stressful, resulting in a marked response of dissatisfaction. Nevertheless, the majority

of learners disagreed with the negative statements concerning inquiry-based learning. Therefore, it can be concluded that learners perceived guided IBL as beneficial to their learning, as evidenced by their overall satisfaction with the inquiry process.

Chapter 5

5.0 Discussion of the results, conclusion and recommendations

5.1 Introduction

The present study investigated the effectiveness of guided Inquiry-Based Learning (guided IBL) in enhancing problem-solving skills among 11th graders in Development Studies (DS). The primary objective was to determine whether guided IBL could significantly improve learners' ability to address complex problems, such as evaluating potential solutions to climate change in Lesotho. This chapter critically discusses the research findings in relation to the existing literature and the research questions outlined below.

Research questions

- \checkmark What is the current level of problem-solving skills among learners in DS?
- ✓ What influence does guided IBL have on learners' problem-solving skills in DS?
- ✓ What is the level of satisfaction among learners on the use of guided inquiry-based learning in DS class?

5.2 The current level of problem-solving skills among learners in DS

The first research question aimed to determine the level of problem-solving skills among learners in Development Studies (DS). To address this question, a pre-test was conducted, as described in the preceding chapter. The findings revealed that learners' problem-solving abilities were significantly deficient prior to the intervention. Primarily, none of the 40 participants in the study achieved a passing mark, with no learner attaining the minimum of 10 out of 20 marks required for a pass. The highest score recorded was 8 out of 20 marks.

These results are consistent with those reported by Oroz et al. (2023), who found that grade 9 chemistry learners also performed poorly on problem-solving tasks during the pre-test phase. Similarly, in the current study, both the control and experimental groups exhibited low scores in

problem-solving assessments. Supporting this, the test of homogeneity for the groups at the pretest level indicated that p = 0.895 > 0.05, suggesting that the performance of learners was comparable between the two groups. Sippl (2021) suggested that without a structured, step-bystep approach like the IDEAL model of problem-solving, teachers struggle to effectively develop students' problem-solving skills.

Furthermore, Wang (2015) argued that traditional learning approaches hinder learners' ability to cultivate essential skills, such as problem-solving and critical thinking. Freire (1970) similarly criticised traditional education methods for failing to support active learner participation, describing them as dehumanising and perpetuating the status quo, where learners are denied epistemological access. This implies that the poor problem-solving performance observed among learners may be linked to the teaching approaches employed. In alignment with this view, Hu (2024) emphasised that traditional learning methods deprive learners of the opportunity to understand concepts deeply and develop vital skills.

While the poor performance in DS may be accredited to the nature of teaching and learning provided, Adeoye et al. (2023) and Marshall (2022) argued that problem-solving is inherently learner-centred and should be taught accordingly. Adeoye (2023) further argued that learners with well-developed problem-solving skills tend to perform better on tasks requiring such abilities. Pratiwi et al. (2021) recommended the IDEAL model as an effective guide for teachers to help learners more easily build problem-solving skills. Additionally, the Lesotho Basic Education Curriculum Policy (LBECP) advocates for an educational approach in Lesotho that emphasises the development of critical thinking and problem-solving skills, highlighting the importance of learner-centred teaching and learning (MoET, 2021).

The discussion now proceeds to the second research question.

5.3 The influence of guided inquiry-based learning in enhancing problem-solving skills of learners in DS

The second research question aimed to evaluate the influence of guided inquiry-based learning (guided IBL) on enhancing problem-solving skills among learners in Development Studies (DS).

To address this question, two phases were utilised: observations during the intervention and a posttest conducted at the end of the intervention. This section discusses the observations made during the intervention with the group that was treated with guided IBL.

The observations from the first lesson, as presented in Table 4.7, indicated that learners were unfamiliar with the process of inquiry. Although they exhibited some interest in the lesson, they struggled to actively participate and pose questions related to the presented problem or topic. The evidence supporting these assertions is drawn from the first statement on the orientation phase, where 75% of the responses were marked as "no," indicating that learners were not well-acquainted with guided inquiry. Similarly, in the conclusion phase, where learners were expected to reflect on the information gathered in previous stages and apply it to the identified problems, all responses were marked as "no." This initial lack of engagement was attributed to learners' limited exposure to teaching approaches such as guided IBL.

At this stage, it was reported that the teacher played a crucial role by providing continuous support to learners during the inquiry process. The IDEAL model of problem-solving, which informed this study, emphasises that teachers can adopt this model to effectively teach problem-solving skills (Annizar et al., 2020). Furthermore, Setyadi and Triyanto (2019) highlighted that during problemsolving, teachers must continuously support learners, particularly in identifying the problem of investigation as the first step. Pedaste et al. (2015) similarly argued that in guided inquiry, teachers should consistently support learners and provide them with necessary learning materials throughout the lesson. Resultantly, it can be concluded that the IDEAL model played an important role in the initial stage of the lesson, as the teacher utilized its components to guide learners through the inquiry process.

The third lesson of the intervention, as shown in Table 4.8, demonstrated significant improvement in learners' behaviour in conducting inquiry-based problem-solving. Learners were reported to have improved in identifying investigation problems and proposing solutions. For instance, in the first lesson, during the orientation phase focused on topic presentation and problem identification, 75% of the responses were marked as "no." However, by the third lesson, 75% of the responses were marked as "ges," suggesting that as learners became accustomed to the inquiry process, their performance and behaviour in solving problems improved significantly.

Supporting this finding, Nisa et al. (2018) observed that senior high school learners in Indonesia gradually improved their performance after being exposed to inquiry-based learning. Similarly, Sukantawaree et al. found that in the Thai context, 4th graders' problem-solving skills improved from 14.28% to 54.14% after exposure to IBL. This suggests that guided IBL has the potential to enhance learners' problem-solving skills. As a product of constructivism, this finding aligns with Andrini's (2016) view that the learner-centred nature of guided IBL allows for active classroom participation, fostering the development of essential skills.

Active participation in the learning process appears to be crucial for developing problem-solving skills. The IDEAL model underscores the importance of learner engagement by allowing them to explore ideas and satisfy their curiosity through interaction with various materials to solve problems (Gusau & Mohamand, 2020). It can thus be concluded that as learners become more familiar with the inquiry process, their problem-solving skills improve compared to their initial encounters.

In the final lesson of the intervention, learners appeared to have developed all the essential characteristics indicative of problem-solving skill development. Markedly, all statements on the observation checklist were marked as "yes." This record indicates that the inquiry process, with the teacher's guidance, significantly enhanced learners' ability to identify problems, develop procedures to address them, explore relevant materials, and apply appropriate solutions to the identified issues.

To support this assertion, Canning and Masika (2022) illuminated that guided IBL encourages active learner participation, enabling them to construct their own knowledge and shape their understanding. Further, Heick (2023) added that engaging learners in question-posing helps them develop effective problem-solving skills and apply these solutions to real-world scenarios. This evidence suggests that guided IBL is a valuable tool for fostering problem-solving skills among learners in DS. As an advocate of scaffolding, Wood et al. (1976) emphasised the importance of continued support during learning, arguing that learners need proper guidance from a more knowledgeable other to grasp difficult concepts. This suggests that in guided IBL, the teacher's support positively impacted learners' inquiry processes, aiding their development of problem-solving skills.

In summary, the observations indicate that when learning is guided and learners are actively involved through question-posing and collaboration, their problem-solving abilities improve. Additionally, when teachers have clear guidelines, such as the IDEAL model for implementing IBL, it becomes easier to assist learners throughout the process.

5.4 The post-test results discussion

This section presents the outcomes of the post-test administered after the completion of the intervention. The purpose of the post-test was to assess the impact of guided Inquiry-Based Learning (guided IBL) on learners' problem-solving skills. Both the interest group, which received guided IBL, and the comparison group, which was taught using traditional methods, were post-tested to identify any differences in performance following the treatment.

The results revealed that learners who were taught using guided IBL demonstrated significantly better performance compared to those who were taught using conventional methods. Specifically, Table 4.10 indicates that the mean score for the experimental group was m=13.35m, whereas the mean for the control group was m=6.15m. Furthermore, a significance level of p=0.00 was recorded, signifying p<0.05, which suggests that there were significant differences between the results of the two groups at the post-test level. This difference is further supported by the higher mean scores of the experimental group compared to the control group.

These findings highlight the superior potential of guided IBL over traditional approaches in enhancing learners' problem-solving skills in Development Studies (DS). Exceptionally, at the pretest level, no significant differences were observed between the two groups, as evidenced by a significance level of p=0.895, which is greater than 0.050.

The results are consistent with previous research by Masilo (2018), Mwenda and Ndayambaje (2021), Nisa et al. (2018), and Sukontawaree et al. (2022), all of which found that IBL significantly improves learners' problem-solving skills and critical thinking abilities compared to traditional methods. Specifically, Mwenda and Ndayambaje (2021) propounded that the components of IBL, such as cooperative and self-directed learning, facilitate out-of-the-box thinking and the development of higher-order thinking skills. This study corroborates that allowing learners to

interact and engage in problem-solving activities enhances their performance in addressing complex questions.

Nachiappan et al. (2018) emphasised that high-order thinking skills, including problem-solving, enable learners to manipulate information and apply solutions effectively. Similarly, Marshall (2021) suggested that problem-solving extends beyond mere memorisation and reproduction of knowledge to include understanding and applying facts to real-world situations. Therefore, effective problem-solving is crucial in the teaching and learning of DS. Traditional learning methods, which tend to inhibit the development of essential skills, should be reconsidered. Freire (1970) further argued that conventional teaching approaches create unjust learning environments that do not foster learners' ability to actively engage and construct their own understanding.

In conclusion, the post-test results indicate that properly guided learners can successfully develop and apply problem-solving skills. According to Bloom (1969), learners who achieve high levels of problem-solving proficiency are capable of synthesizing, appraising, analysing, and assessing situations. This study demonstrates that guided IBL can effectively stimulate learners' curiosity through questioning, exploration, and knowledge construction, leading to enhanced problemsolving skills. Thus, it is recommended that DS teachers consider adopting guided IBL as a pedagogical approach to support learners in acquiring these crucial skills. The next section will discuss learners' satisfaction with the use of guided IBL in DS classrooms.

5.5 The level of satisfaction among learners on the use of guided inquiry-based learning in DS class

In an epistemologically accessible and equitable learning environment, the feelings and satisfaction of learners are highly valued (Andrini, 2016; Freire, 1970; Heick, 2023). Accordingly, the final research question of this study aimed to assess learners' satisfaction with the use of guided Inquiry-Based Learning (guided IBL) in their Development Studies (DS) classrooms. Despite its novelty, learners reported a high level of satisfaction with guided IBL when evaluated on a five-point Likert scale, particularly in relation to developing problem-solving skills. Specifically, as detailed in Table 4.11, a substantial number of learners expressed that guided IBL was an effective approach for enhancing their problem-solving skills, with 60% indicating that it facilitated their understanding of difficult concepts in DS.

Moreover, learners greatly appreciated the opportunities to ask questions and collaborate during the inquiry process. The IDEAL model of problem-solving supports the notion that allowing learners to ask questions contributes to a sense of belonging within the classroom. Mackenzie (2016) emphasised that IBL promotes social interaction among learners, which aids them in addressing various problems. Similarly, Vygotsky (1978) highlights the significance of social interaction in the classroom, arguing that it enables novice learners to receive guidance from more knowledgeable peers. This implies that facilitating interaction and inquiry enhances learners' problem-solving abilities.

Prominent advocates of learner-centered teaching, including Brunner (1978), Freire (1970), and Vygotsky (1978), underscore the necessity of ongoing support during the learning process. They argue that continuous guidance improves learners' ability to perform tasks and complete activities independently. The IDEAL model, as articulated by Bransford and Stein (1984), insists that effective problem-solving instruction requires learners to receive assistance in identifying problems from the outset to successfully complete the problem-solving process. In this study, learners reported meaningful satisfaction with the inquiry process, noting that it enabled them to identify relevant solutions to problems.

Nevertheless, despite these positive findings, the literature reveals ongoing challenges in implementing IBL in various contexts. For instance, Simbarashe and Gherda (2019) identified that in Zimbabwe, the implementation of IBL is hindered by the demands of public examinations, which limit the time available for such approaches. Similarly, Koekoe (2023) acknowledged that while teachers in Lesotho hold a positive attitude towards IBL, they often lack the necessary competencies and skills for its effective classroom implementation.

To address these constraints, several scholars propose potential solutions. Koekoe (2023) and Pesqueira (2020) suggest that the responsible ministries should provide ongoing teacher training through workshops or seminars focused on IBL implementation. Furthermore, Heick (2023) advocates for collaborative efforts among teachers to share skills and strategies for conducting inquiry-based learning. Regarding public examination results, Kuykendall (2022) proposes that teachers align inquiry procedures with the syllabus's end-of-level objectives to better integrate IBL within the assessment framework.

In summary, while learners expressed satisfaction with guided IBL and its benefits for problemsolving skills, challenges in its implementation persist across various contexts. Addressing these challenges through targeted training and alignment with assessment objectives could enhance the effectiveness and adoption of IBL in diverse educational settings

5.6 Conclusion

This study investigated the impact of guided Inquiry-Based Learning (guided IBL) on enhancing problem-solving skills among Grade 11 learners in Development Studies (DS) at a high school in Maseru. Employing a quantitative quasi-experimental design, the study included two groups: an experimental group that experienced guided IBL and a comparison group that was taught using conventional methods. Initially, pre-test results showed no significant differences between the two groups, establishing a baseline for comparison. However, post-test results demonstrated that the experimental group, which underwent guided IBL, exhibited substantial improvement in problem-solving skills compared to the control group. Additionally, classroom observations revealed that learners' understanding of concepts, collaboration skills, information retrieval, and problem-solving application meaningfully improved as they adapted to the inquiry process.

The findings suggest that guided IBL has considerable potential for fostering a deeper understanding and active engagement with learning materials, thereby enhancing learners' problem-solving abilities. Feedback collected via a Likert scale reflected high levels of satisfaction among learners with the guided IBL approach. The results underscore the transformative potential of guided IBL in educational practices within DS teaching and learning. Furthermore, the IDEAL model of problem-solving provided a clear, step-by-step framework for implementing IBL phases, which was highly valued for its simplicity and effectiveness in teaching problem-solving skills.

In conclusion, the study affirms that guided IBL positively impacts problem-solving skills among Grade 11 learners in DS. The results indicate that learners benefit more from active inquiry compared to passive learning environments. Active learning approaches, therefore, emerge as more effective in fostering problem-solving skills.

5.7 Recommendations

Recommendations in research are crucial, as they offer guidance and advice from the researcher to various stakeholders (Nair, 2024). Such recommendations typically stem from the findings of the current investigation. Thus, this study proposes the following recommendations:

Implementation of Guided Inquiry-Based Learning (IBL) in Lesotho's Education Context

- Incorporation of IBL in Curricula: The literature review indicates that teachers generally do not harbour negative attitudes toward IBL. Therefore, it is recommended that schools in Lesotho integrate this teaching approach into their curricula, particularly in Development Studies (DS). This recommendation is supported by the Curriculum and Assessment Policy Statement (CAPS) and the Lesotho Basic Education Curriculum Policy (LBECP), which emphasize that education in Lesotho should aim to enhance learners' abilities to tackle real-life situations (MoET, 2018; MoET, 2021).
- 2. **Professional Development for Teachers:** Echoing the suggestions of Koekoe (2023), it is recommended that the Ministry of Education and Training implement on-going professional development programs for teachers. These programs should focus on equipping educators with the skills and knowledge necessary to implement 21st-century teaching methods such as guided IBL. Professional development could include workshops, seminars, and guided group discussions. Additionally, this study advocates that the Ministry encourage collaborative efforts among teachers. By sharing ideas and experiences, teachers can better implement IBL and enhance their teaching practices.
- 3. Engaging Learners in Active Learning: It is advisable for teachers to engage learners in active learning processes. When introducing new approaches, such as guided IBL, active engagement will help learners relate more easily to the new methods. In DS, the theme "education and development" encompasses topics such as pollution, resources, and climate change. Allowing learners to construct their own knowledge on these topics can significantly boost their classroom engagement.

5.8 Limitations of the Study

Despite successfully demonstrating the effectiveness of guided IBL in improving problem-solving skills among DS learners, this study acknowledges certain limitations. Primarily, the focus on problem-solving skills was a limitation, as it excluded other essential skills such as critical thinking, assessment, communication, and leadership, which are also vital in DS education. Furthermore, financial constraints and limited research time restricted the study's scope.

5.9 Areas for Further Research

Further research should look into the long-term impacts of guided IBL across various grade levels and subjects. Additionally, this study's results provide insights into conducting inquiry-based learning in the classroom with a focus on problem-solving skills. Subsequent research could explore how guided IBL can enhance other critical skills, such as creativity, critical thinking, communication, and computational skills. Furthermore, examining different types of IBL across various subjects could provide valuable insights into their impact on teaching and learning.

5.10 Reflections

Reflecting on this research journey reveals several insights into the use of guided IBL in DS classrooms. Importantly, learners excel in problem-solving when they actively participate and engage in the learning process. The enthusiasm observed during the guided IBL intervention accentuates the importance of creating a conducive learning environment that motivates learners to explore ideas independently.

Despite facing financial and health challenges, this study reaffirms the value of innovative teaching approaches tailored to the diverse needs of learners while fostering essential skills like problemsolving. In conclusion, this research offers valuable contributions to understanding effective pedagogical strategies for developing 21st-century skills among learners.

REFERENCES

- Abiodun-Oyebanji, O. J. (2017). Research variables: Types, uses and definition of terms. In Book: Research in education (pp43-54) His Lineage Publishing House
- Adeoye, M. A. & Jimoh, H. A. (2023). Problem-Solving among 21st-Century Learners towards Creativity. *Thinking Skills and Creativity Journal*, 6(1). <u>https://doi.org./1023887/tscj.v6i1.62708</u>
- Ahuja, H. (2021). Why Developing Problem-solving Skills Important? <u>https://elearningindustry.com/6-stratergies-to-instill-solving-skills-in-learners</u>
- Ali, E. (2024). Re: What is Intact Sampling? https://www.researchgate.net/post/what-is-intact-sampling/
- Alharahsheh, H. H. & Pius, A. (2020). A review of key paradigms: Positivism vs interpretivism global. *Academic journal of humanities and social science*, 2(3), 39-43. <u>https://gairc.com.gajhss</u>
- Amalia, E. (2018). Collaborative Learning: The Concepts and Practices in the Classroom. https://doi.10.31219/osf.oi/xn67t
- Andrini, V. S. (2016). The Effectiveness of Inquiry Learning Method to Enhance Students' Learning Outcome: A Theoretical and Empirical Review. *Journal of Education and Practice*,7(3), 38-42.
- Annizar, A. M. (2018). Problem-Solving using IDEAL Model Judging from the Conceptual Tempo Cognitive Style at Ambulu High School. Thesis: State University of Malang
- Annizar, A. M., Apriyono, F., Jakaria, M. H. D., Masrurotullaily & Mukhlis, M. (2020). Problem-Solving of Rational Inequality based on IDEAL Model. *Journal of Physics: Conference Series* <u>https://doi.10.1088/1742-6596/1465/1/012033</u>
- Armstrong, P. (2010). Bloom's Taxonomy. https://cft.vanderbilt.edu/guides-sub-pages/booms/taxonomy/
- Arocena, R. (2016). Development Studies as an Interdisciplinary Field: Research, Teaching and Interdisciplinary Building in Uruguary. *Issues in Interdisciplinary Studies*, *34*(2), 164-182.
- Aydin, G. (2020). The Effects of Guided Inquiry-Based Learning Implementations on 4th Grades Students and Elementary Teacher; A case Study. *Ilogretim Online-Elementary Education Online*, 19(3) 1155-1184. <u>https://doi.17051/ilkoline.2020.727298</u>

- Bagwasi, M. M. (2017). A Critique of Botswana s' Language Policy from Translanguaging Perspective. *Current Issues in Language Planning, 18*(2), 199-214. <u>https://doi.org/10.1080/14664208.2016.1246840</u>
- Bagwasi, M. M. (2018). The Major Educational Policies, Models and Ideas that have Influenced Botswana Education System. *Policy Futures in Education*, 17(3). https://doi.org/10.1177/1478210318807779
- Bartschi, G. J. (2024). What are Demographics? Examples, Definitions and Step by Step Guide. https://www.supersurvey.com/demographic-information
- Berger, R. (2020). Here's What's Wrong with Bloom's Taxonomy: A Deeper Learning Perspective. <u>https://www.edweek.org/education/opinion-heres-whats-wrong-with-blooms-taxonomy-a-</u> <u>deeper-learning-perspective</u>
- Berma, M & Sulehan, J. (2004). Being Multi-Disciplinary in Development Studies: Why and How Adekamika 64 (Januari), *Seminar on Directions and Destinations in Development Studies*, 43-63.
- Bevans, R. (2023). Understanding Confidence Intervals: Easy W=Examples and Formulas https://www.scribbr.com/statistical confidence/interval/
- Beyrow, M., Godau, M., Heidmann, F., Langer, C., Wettach, R. & Mieg, H. A. (2019). Inquiry-Based Learning in Design. <u>https://doi.1007/978-3-030-14223-0-22</u>
- Bhandari, P. (2023). Independent vs. Dependent Variable: Definition and Examples. https://www.scribbr.com/methodology/indeppendent-and-dependent-variables/
- Bhandari, P. (2023). Population vs. Sample; Definitions, Differences and Examples. https://www.scribbr.com/methodologylog/population-vs-sample
- Bhandari, P. (2023). What is Quantitative Research? Definition, Uses and Methods. <u>https://www.scribbr.com/</u>
- Bhosale, U. (2024). Setting Limits and Focusing on your Study: Exploring Scope and Delimitation https://www.enago.com/academy/scope-and-delimitations/
- Bloom, B. S (1956). Taxonomy of Educational Objectives. Vol.1: Cognitive Domain. New York: Mckay, 20, 24.

- Bloom, B. S. (1969). *Taxonomy of Educational Objectives: The classification of educational goals*: Handbook I, cognitive domain.
- Boikhutso, K. (1993). Vocationalisation of the School Curriculum in Botswana. Mid Unpublished Dissertation, Bristol, University of Bristol.
- Boikhutso, K. (2013) How well does Botswana s' Social Studies Curriculum Articulate Gender Issues? A Preliminary Overview. *Journal of Curriculum and Teaching* 2(1) <u>https://dx.doi.org/10.5430.jct.v2nlp54</u>
- Bransford, J. D. & Stein, B. S. (1984). *The IDEAL problem solver: A guide for improving thinking, learning, and creativity.* New York, NY: W. H. Freeman
- Brown, L. & Lara V. (2011). Professional Development Module on Collaborative Learning. El Paso Community College. <u>https://www.texascollaborative.org?collaborative-learning-module</u>
- Bruner, J. S. (1961). The Act of Discovery. Havard Educational Review 31, 21-32
- Bruner, J. S. (1966). Toward a theory of instruction. Cambridge, Harvard University Press
- Bruner, J. S. (1978). *The Role of Dialogue in Language Acquisition*. In A. Sinclair, R. J. Jarvelle, and W.J. M Levelt (eds) *The Child's Concept of Language*. New York: Springer-Verlag.
- Budert-Waltz, K & Levitas, M. (2023). Pre-test-post-test design: definition, types & examples. https://study.com>learn>lesson
- Brydges, C. (2019). Effect Size Guidelines, Calculations, and Statistical Power in Gerontology. https://doi:10.1093/geronb/gbz033
- Bybee, R. W. (2006). Scientific inquiry and science teaching. In L. B. Flick, & N. G. Lederman (Eds.), Scientific inquiry and nature of science: implications for teaching, learning and teacher education. Dordecht: Springer 1-14
- Campbell, M. J. & Shantikumar (2016). Parametric & Non-Parametric Tests Comparing Two Groups or More. Faculty of Public Health. <u>https://www.healthknowledge.org.uk/public-health-textbook/resources/methods</u>
- Campbell, M. J. & Swinscow T. D. V. (2009). Statistics at Square One 11th ed. Wiley-Blackwell: BMT Books.

- Canning, J. & Masika R. (2022). The Scholarship of Teaching and Learning (SoTL): The Thorn in the Flesh of Educational Research Studies in Higher Education Brighton.ac.uk.
- Carroll, K. & Goodfriend W. (2023). Validity & Reliability in Research; Definition and Importance. <u>https://study.com/academy/lesson/validity-and-reliability-how-to-assess-the-quality-of-research-study.html</u>
- Chia, J. (2023). 7 Reasons Why Data Analysis is Important for Research. https://justjooz.com
- Chris, D. (2023). The 5 phases of inquiry-based learning: helpful professor. <u>https://helpful-professor.com/inquiry-based-learning/</u>
- Clark, D. (2015). Bloom's Taxonomy: The Affective Domain. https://www.nwlink.com/donclark/hrd/Bloom/affective_domain.html
- Cohen, J., Cohen, P., West, S. G & Aiken, L. S. (2013). *Applied Multiple Regression Correlation Analysis* for the Behavioural Science (3rd ed.) Routledge Taylor & Francic Group, LLC.
- Cohen, L., Manion, L. & Morrison, K. (2018). *Research methods in education*. Routledge. Tyler & Francis Group.
- Cohen, J. (1992). Tutorials in Quantitative Methods for Psychology: *Psychological Bulletin*, 112 (1). https://doi:10.1037/0033-2909.112.1.155
- Connell, L. (2012). The Impact of Traditional SVS. Interactive Instruction on Students' Comprehension. Education Masters' Paper 203. <u>https://fisherpub.sjf.edu/education-ETD-masters/203</u>
- Connelly, L. M. (2013). Demographic Data in Research Studies. Medsurg Nursing. https://link.gale.com/apps/doc/A3416870/AONE
- Creswell, J. W. & Creswell, J. D. (2018). *Research Design: Qualitative, Quantitative and Mixed Methods Approaches.* Sage Publications.
- Creswell, J. W. & Plano Clark V. L. (2011). *Designing and conducting mixed methods research*. 2nd edition. Thousand Oaks: Sage.
- Currie-Alder, B. (2016). The State of Development Studies: Origins, Evolution and Prospects. *Canadian Journal of Development Studies*, 37(1), 5-26. <u>https://dx.doi.org/10.1080/02255189.2016.1135788</u>

- Dambudzo, I. I. (2015). Curriculum Issues: Teaching and Learning foe Sustainable Development in Developing Countries: Zimbabwe Case Study. Journal of Education and Learning, 4(1), <u>https://dx.doi.org/10.5539/jel.v4n1p11</u>
- Dewey, J. (1986). Experience and Education Forum. Tyler & Francis Group, 241-252.
- Dewey, J. (1966). Democracy and Education: An Introduction to the Philosophy of Education. New York: Free Press.
- Davrik, R., Pilten, P & Tas, A. M. (2020). Effect of Inquiry-Based Learning Method Supported by Metacognitive Strategies on Fourth-Grade Students' Problem-Solving and Problem-Posing Skills: A Mixed Methods Research. *International Electronic Journal of Elementary Education*, 13(2), 287-308. https://doi.org/10.26822/iejee.2021.191
- Examination council of Lesotho Markers' Report: Development Studies (2023). https://www.examscouncil.org.ls/examiners-report
- Examination Council of Lesotho Markers' Report: Development Studies (2022) https://www.examscouncil.org.ls/examiners-report.
- Examination Council of Lesotho Markers' Report: Development Studies (2021). https://www.examscouncil.org.ls/examiners-report
- Examination Council of Lesotho Markers' Report: Development Studies (2020). https://www.examscouncil.org.ls/examiners-report
- Examination Council of Lesotho Markers' Report: Development Studies (2019). https://www.examscouncil.org.ls/examiners-report
- Eke, C. (2022) Differences between empirical review: Accademic guide Retrieved from https://writerskig.com
- Ernst, D. C., Hodge, A. & Yoshinobu, S. (2017). What is Inquiry-Based Learning? *Notices of the American Mathematical Society* 64(06), 570-574. <u>https://doi.1090/noti1536</u>
- Field, A. (2018). Discovering Statistics Using IBM SPSS Statistics. Sage Publications.
- Field, A., & Miles, J. (2012). Discovering Statistics Using IBM SPSS Statistics. SAGE Publications Ltd.

- Fissore C., Marchisio, M., Roman, F. & Sacchet, M. (2021). *Development of problem-solving skills with maple in higher education In*: Corless, R. M.
- Fitzgerald, M., Danaia, L. & Mckinnon, D. H. (2019). Barriers Inhibiting Inquiry-Based Science Teaching and Potential Solutions: Perceptions of Positively Inclined Early Adopters. *Res Sci Edu*. <u>https://doi.org?10.1007?s11165-017-9623-5</u>
- Ford, C. (2023). Getting Started with Analysis of Covariance. https://library.virginia.edu/data/articles/getting-started-with-analysis-of-covariance/
- Freire, P. (1970). Pedagogy of the oppressed. New York: Seabury Press.
- Fuke, J. (2017). *The nature of problem-solving: Using research to inspire 21st century learning*. OECD Publishing. Paris.
- Gates, S. (2018). Benefits of Collaboration. <u>https://www.nea.org?professional-excellence/student-engagement/tools-tips/benefits-to-collaboration</u>
- Gholam, A. (2019). Inquiry-Based Learning: Student Teacher's Challenges and Perceptions: *Journal of Inquiry and Action in Education* 10(2).
- Gillies, R. M. (2023). Teaching Science that is Inquiry-Based: Practices and Principles. In: Thomas, G.P., Boon, H. J (eds) Challanges in Science Education. Palgrace Macmillian, Chan.https://doi.org/10.1007/978-3-031-1809-7-3
- Golder, J. (2018). Constructivism: A Paradigm for Teaching and Learning. International Journal of Research and Analytical Reviews 5(3)
- Gunawan, G., Harjono, A., Nisyah, M., Kusdiastuti, M. & Herayanti, L. (2020). Improving Students' Problem-Solving Skills using Inquiry Learning Model Combined with Advance. *International Journal of Instruction*, 13(4), 427-442. <u>https://doi.org?10.29333?iji.2020.134279</u>
- Gusau, N. M. B. & Mohamad, M. M. (2020). Problem-Solving Skills based on IDEAL Model in Implementing Undergraduate Final Year Project. *Journal of Technology and Humanities*. 1(1). <u>https://doi.org/10.53797/jthkkss.v1i.4.2020</u>
- Hala, M & Xhomara, N. (2022). The Impact of Inquiry-Based Learning on Problem-Solving Skills and Conceptual Knowledge Building. *Psychology and Education*, 59(2), 909-921

- Hayes, A. (2024). What is a Confidence Interval and how do you Calculate it? https://www.investopedia.com/terms/confidenceinterval/
- Heick, T. (2023). Effective Teaching Strategies for Inquiry-Based Learning. https://www.teachthought.com/pedagogy/inquiry-based-teaching-strategies/
- Helmenstine, T. (2024). Difference between Independent and Dependent Variables. https://www.thoughtco.com>independent>variable>dependent>variable
- Hettne, B. (1990). *Development theory and the three worlds*. Harlow: Longman scientific and technical. https://doi.org/10.1177/030913259201600121
- Howell, D. C. (2017). Statistical Methods for Psychology Cengage Learning, 43(1). https://doi:10.2307/2348956
- Iulian F. (2018). Understanding Inquiry, An Inquiry into Understanding Conception of Inquiry-Based Learning in Mathematics. Masters Theses, Concordia University.
- Hu, J. (2024). The Challenge of Traditional Teaching Approach: A Study on the Path to Improve Classroom Teaching Effectiveness Based on Secondary Schools' Students Psychology. *Proceedings of the 2nd International Conference on Social Psychology and Humanity Studies* 50(1) 213-219. <u>https://doi.10.542541/2753-7048-50/</u>
- Jansen, D. (2023). Research Philosophy and Paradigms: Positivism, Interpretivism & Pragmatism Explained Simply. <u>https://gradcoach.com/research-philosophy/</u>
- Kalplan, S. (2024). How to make a Frequency Table: Boost your Skills Now. https://enjoymachinelearning.com/blog/how-to-do-8a-frequency-table/
- Koekoe, T (2023). Geography Teachers' Perceptions on the use of Inquiry-Based Learning: A Case of Five High Schools in Mohale-Hoek District; Masters Thesis, National University of Lesotho. <u>https://resipository.tml.nul.ls/handle/20.500.14155/2017?show=full</u>
- Komba, A., Kontinen, T & Msoka, C (2019) Development Studies in Tanzania: Historical Trajectory and Future Visions. In: Baud, I., Basile, E., Kontinen T., Von Itter, S. (eds) Building Development Studies for New Millennium. EADI Global Development Series. Palgrave Macmillian, Cham. <u>https://doi.org/10.1007/978-3-030-04052-9-14</u>

- Kontinen, T., Kilonzo., Msoka, C. (2022). Hybrid Identities of Development Studies in Tanzania: *Researches in Sociology and Anthropology*. <u>https://doi.org/10.4000/rsa.5664</u>
- Kori, K. (2021). Inquiry- Learning in Higher Education: In: Vaz de Carvalho C., Bauters, M. (EDS)
 Technology. Springer, Singapore <u>https://doi.org?10.1007/978-981-16-2082-94</u>
- Kramer, T. J., Zeccardi, J., Emhaff, C. W., Williams, C. Dunn, R. J & Rose J. (2022). How Timing and Authority in Peer Review Impact STEM Students: A Comparative Assessment of Writting and Critical Thinking in Kinessiology Courses. https://www.semanticschol/.ar.org/paper/37755563ea971465b33
- Kuykendall, K. (2022). 4 Common Obstacles to Implementing Inquiry-Based Learning and How to Overcome them. <u>https://www.edutopia.org/article/4-common-obstacles-implementing-inquiry-based-learning-and-how-overcome-them/</u>
- Kulkarni, H. V. & Patil, K. P. (2022). A Uniformly Superior Exact Multi-Sample Test Procedure for Homogeneity of Variances Under Location-Scale Family of Distributions <u>https://www.researchgate.net?publication?360107508-A-test-for-homogenity-of-variances</u>
- Laal, M. & Ghodsi, S. M. (2012). Benefits of Collaborative learning 486-490. https://doi.org?10.1016/isbspro.2011.12.091
- Lekhanya, N. D. & Raselimo, M. G. (2022). Exploring Alignment Between the Lesotho Secondary DS Teachers' Pedagogical Practices and Work Related Skills. *European Journal of Teaching and Education*, 4(3): 13-27. <u>https://doi.org/10.33422/ejte.v4i3.737</u>
- Lekhanya, N. D (2020). Exploring Alignment between DS Teachers' Pedagogical Practices and Work-Related Competencies in Selected High Schools of Lesotho; MAEd Dissertation. The National University of Lesotho, Roma.
- Lemov, D. (2017). Bloom's that-Pyramid is a problem. https://www.teachlikeachampion.com/blog/blooms-taxonomy-pyramid-problem/
- Lincoln, Y. (1985). The Making of a Constructivist; A Remembrance of Transformations Past. In the Paradigm Dialog. Edited by Egon G. Cuba Newburry Park: Sage 67-87

- Lincoln, Y., Susan A. L. & Egon, G. Cuba (2011). *Paradigms and Perspectives in Contention*. In The Sage Handbook of Qualitative Research.
- Lippmann, M. (2022). Inquiry-Based Learning in Psychology. In: Zumbach, J. Bernstein., D., Narciss, S. Marsico, G (eds) International Handbooks of Education. Springer, Cham. <u>https://doi.org/10.1007?978-3-030.26248-8-59-3</u>
- Luft, J. A., Jeong, S., Idsardi, R. & Gardner, G. (2022). Literature Review, Theoretical Framework and Conceptual Framework: An Introduction for New Biology Education Researchers. Life Education 21(3). <u>https://www.doi.org/10.1187/cbe21-05-0134</u>
- Mackenzie, T. (2016). *Dive into Inquiry: Aplify Learning and Empower Students' Voice*. California: EdTechTeam Press.
- Macleod, S. (2024). Jerome Bruner's Theory of Learning and Cognitive Development. https://www.simplypsychology.org/bruner.htral
- Mandaar, P. S. & Vijayakumar, B. (2020). Theoretical Foundations of Design Thinking- A Constructivism Learning Approach to Design Thinking. *Thinking Skills and Creativity*, 36. https://doi.org/10.1016/j.tsc.2020.1000637
- Mandina, S. (2024). Teachers' Perceptions and Implementation of IBL IN Rural Schools. *African Journal of Chemical Education*, 14 (1). <u>https://www.ajol.info/index.php/ajce/article/view/262757</u>
- Marshall, M. (2022). Benefits of problem-solving in the K-12 Classroom. https://www.competitionscience.org
- Marshall, J. C. (2013). Succeeding with Inquiry in Science and Math Classrooms. ASCD. https://www.amazon.com/succeeding-inquiry-science-math-classrooms/dp/141661608x
- Martin, A.J., Collie, R. J., Stephan, M., Flesken, A., Halcrow, F. & McCourt, B. (2024). The Role of Teaching Support in Assisting Students' Transition to High School, Learning and Individual Differences. 109. <u>https://doi.org.10.1016/j.lindif.2023.102382</u>
- Mark, T. (2020) The Statistical Analysis Using SPSS by Titocan Mark. YouTube Video. https://www.youtube.com

- Masilo, M. M. (2018). *Implementing Inquiry-Based Learning to Enhance Grade 11 Students' Problem-Solving Skills in Euclidean Geometry*. Doctorate Dissertation: University of South Africa.
- Maxwell, J. A. & Delaney, H. D. (2018). *Designing Education and Social Research: Principles and Procedures.* Sage Publications
- McCombes, S. (2023). Sampling Mathods: Types, Techniques & Examples. https://www.scribbr.com/methodology/sampling-methods/
- Menon, K. (2024). Understanding the fundamentals of the Confidence Interval in Statistics. https://www.simplilearn.com/tutorials/data-analytics-tutorial/confidence-intervals-in-statistics
- Middleton, F. (2023). The 4 Tyoes of Validity in Research: Definitions & Examples. https://www.scribbr.com/author/fionamiddleton/
- Ministry of Education (2016). Kurikulum Standard of Prasekolah Kebangsaan Putrajava: Bahagian Pembangunan Kurikulum
- Ministry of Education, Sports and Culture (1982). The Education Sector Survey: Report of the Task Force Maseru: Ministry of Education, Sports and Culture
- MoET (2021). Lesotho Basic Education Curriculum Policy: Ministry of Education and Training Docs
- MoET (2020). Development Studies Syllabus: Ministry of Education and Training Docs
- MoET (2018). Lesotho General Certificate of Secondary Education, Development Studies Syllabus: Ministry of Education and Training Docs.
- MoET (2009). Curriculum and Assessment Policy: Ministry of Education and Training Docs.
- Mudavanhu, Y. (2017). Quality of Literature Review and Discussion of findings in Selected Papers on Intergration of ICT in Teaching, Role of Mentors, and Teaching Science Through Science, Technology, Engineering, and Mathematics (STEM). Educational Research and Reviews 12(4) 189-201 <u>https://doi.10.5897/ERR2016.3088</u>
- Mwenda, K. S. & Ndayambaje (2021). Effects of Inquiry-Based Teaching on Students' Academic Achievement in Biology in Lower Secondary Schools in Ilaha-Dar es Salaam Tanzania. A Journal of Contemporary Research 18(4) 2-14 <u>https://creativecommons.org/licenses/by-nc-nd/4.0</u>

- Nachiappan, S., Damahuri, A. A., Ganaprakasam, C. & Suffian, S. (2018). Application of Higher Order Thinking Skills (HOT) in Teaching and Learning Trough Communication Component and Spiritual, Attitudes and Values Components in Pre-School. *International Journal of Early Childhood Education Care*
- Nair, A. (2024). Research Recommendations-Guiding Policy-Makers for Evidence-Based Decision Making: Enago Academy <u>https://www.enago.com/recommendation-in-research/</u>
- New Education Policy (2020). 21st Century Skills: Why is Problem-Solving the Need of the Hour for School Students? <u>https://hcljigsaw.com/21st-century-skills-why-is-is-problem-solving-the-need-of-the-hour--for-school-students/</u>
- Nickols, F. (2020). Thirteen Problem-Solving Models https://www.mindtools.com.pages/article/newCT92.htm
- Nisa, E., Koestiari, T., Habibbulloh, M & Jatmiko, B (2018) Effectiveness of Guided Inquiry-Based Learning Model to improve Students' Critical Thinking Skills at Senior High School, *Journal of Physics Conference Series*, 997(1). https://doi.10.1088/1742-6596/997/1/012049

National-Strategic-Plan 2018/19-2022/2023 of Lesotho https://www.gov.ls/nsdp-ii-statergic-focus /

- Nwankwo, J. I. & Emunemu, B. O. (2014). A Handbook on Research In Education and the Social Sciences. Ibadan: Giraffe Books.
- Orosz, G., Nemeth, V., Koracs, L., Somogyi, Z & Korom, E. (2023). Guided Inquiry-Based Learning in Secondary School Chemistry Classes: A Case Study. Chemistry Education Research and Practice, 24, 50-70. <u>https://doi.10.1039/d2r0olla</u>
- Pallant, J. (2016). SPSS Survival Manual: A Step by Stepo Guide to Data Analysis Using IBM SPSS. McGraw-Hill Education (UK).
- Palupi, B. S., Subiyantoro., Rukayah & Triyanto (2020). The Effectiveness of Guided InquiryLearning (GIL) and Problem-Based Learning for Writting Skills. International Journal, 13(1), 713-730 <u>https://doi.org/10.29333/iji.2020.131469</u>
- Pappas, C. (2014). Instructional Design Models & Theories: IBL Model https://elearningindustry.com>Inquir-Based-Learning-Model

- Park, Y. S., Konge, L. & Artino, A. (2019). The Positivism Paradigm of Research 95(1) https://doi:10.1097/ACM.000000000003093
- Patton, M. Q. (2015). Qualitative Evaluation and Research Methods. Thousands Oaks, CA: Sage.
- Pedaste, M., Maeots, M., Siiman, L., Jong, T., Riesen, S. A. N., Kamp, E. T., Manoli, C.C., Zacharia, Z. C. & Tsourlidaki, E. (2015). Phases of Inquiry-Based Learning: Definitions and the Inquiry Cycle. *Educational Research Journal* 14(1). https://doi.10.1016/j.edure.2015.02.003
- Pesqueira, J. (2020). Factors Inhibiting the Implementation of Inquiry-Based Learning and Problem-based Learning in the K-12 New York State Social Studies Classroom. Thesis, Saint Johns University Jamica New York <u>https://scholar.st.johns.edu/thesis-dissertation/83/</u>
- Pratiwi, G., Sujadi, I. & Slamet, I. (2021). Mathematics Problem Solving Ability: An Analysis Based on Ideal Problem Solving Viewed from Learning Style. *International Conference of Mathematics and Mathematics Education*. <u>https://doi:10.2991/assehr.k211122.014</u>
- Rahman, M. (2019). 21st Century Skills "Problem Solving" Defining the Concept, Asian Journal of Interdisciplinary Research 2(1), 71-81. <u>https://doi:10.34256/ajir1917</u>
- Rajeki, S. (2017). Inquiry-Based Language Learning (IBLL): Theoretical and Practical Views in English Classroom: Journal of English Language and Education, 1(2), 135. <u>https://doi:10.29420/ef.v1i2.154</u>
- Ralebese, M. D., Jita, L. C & Chimbi, G. T (2022). "Underprepared" Principals Leading Curriculum Reform in Lesotho. Research in Educational Administration & Leadership, 7(4), 861-897.
 <u>https://doi.org/10.30828.real.1104537</u>
- Raselimo, M & Mahao M. (2015). The Lesotho Curriculum and Assessment Policy: Opportunities and Threats. *South African Journal of Education*, 35(1), 1-12. <u>https://doi.org/10.15700/201503070025</u>
- Rhodes, J. (2023). Frequency: Definition, Measurement and Importance for Data Analysis. https://senioritis.io/mathematics/probability/frequency-distribution-measurement
- Rodriguez, S., Allen, K., Harron, J & Qadri, S. A. (2019). Making and the 5E Learning Cycle: Science Teacher, 86(5), 48-55. <u>https://doi:10.2505/tst18-086-05-48</u>

- Saeidi, M. & Ahmadi, H. (2016). The Effects of Watching Videos in Pre-Reading on EFL Learners' Reading Comprehension and Attitudes. *TESOL International Journal*, 11(2), 15-44. <u>https://www.tesol-international-journal.com/</u>
- Sarah, V. (2022). What is Theoretical Framework? Guide to Organising. https://www.scribbr.com
- Sarbah, B. J. (2020). Constructivism Learning Approaches. https://doi.org/10.13140/RG.2.2.28/38.34241
- Scholl, A. (2023). What is "Inquiry-Based Learning: Types, Benefits, Examples. https://www.splashlearn.com>blog>what-is-inquiry-based-learning
- Schreiber-Barsch, S., Bulungu, J, B &Eble, L. (2023). Encountering Works by Nyerere and Freire: Exploring the Connections between Education for Liberation and Education for Self-reliance in Contemporary Radical Popular Education: *European Journal for Research on the Education and Learning of Adults*, 14(1) 79-96. https://doi.org/10.3384/rela.2000-74.26.4420
- Serin, H. (2018). A Comparison of Teacher-Centered Approaches in Settings. International Journal of Social Sciences & Educational Studies 5(1), 164-167. <u>https://doi.10.23918/ijsses.v5ilp164</u>
- Setyadi, T. Y & Mardiyana, T. (2019). Mathematical Problem-Solving Skills Using IDEAL Model Based on Personality Type. The Second International Conference on Science, Mathematics, Environment and Education https://doi.org?10.1060?1.5139847
- Shabatura, J. (2022). Using Blooms' Taxonomy to Write Effective Learning Outcomes: Assignment and Measuring Students' Learning. <u>https://tips.uark.edu/using-blooms-taxonomy/</u>
- Shieh, G. (2020). Power Analysis and Sample Size Planning in ANCOVA Designs. *Psychometrika*, 85 (1), 101-120. https://doi.org/10.1007/s11336-019-09692-3
- Simbarashe, M. & Gherda, F. J. (2019). The Zimbabwen Advanced Level Physics Teachers Views on Inquiry-Based Physics Intervention and Level of Actual Classroom Practices: International Journal of Research and Innovation in Social Science, 3(6).
- Singh, A. (2023). Blended Learning vs. Traditional Learning: Adetailed Overview of the Two Approaches. <u>https://elearningindustry.com/blended-learning-vstraditional-learning-a-detailed-overview-of-the-two-approaches</u>

- Sippl, A. (2021). Teaching the IDEAL Problem-Solving Method to Diverse Learners: Life Skills Advocate. <u>https://lifeskillsadvocate.com>blog</u>
- Sivakumar, R. (2018). Methods and Resources in Teaching Social Studies: *Journal of Contemporary Educational Research and Innovations*, 8(2), 207-216. <u>https://www.j.ceri.com</u>
- Spadafora,N. & Downes, T. (2020). Scaffolding in Learning In Book: Encyclopedia of Evolutionary Psychological Science. https://www.doi:10/007/978-3-319-16999-6-1350-1
- Sridhar, M. S. (2020) Importance of Literature Review in Research. https://doi.10.13140/RG.2.2.15347.14885
- Stapor, K. (2020). Descriptive Statistics and Inferential Statistics. <u>https://doi:10.1007/978-3-030-45799-</u> 0-2
- Stratton, S. J. (2019). Quasi-Experimental Design (pre-Test and Post-Test Studies) in Prehospital and Disaster Research. *Prehospital and Disaster Medicine*, 34(6), 573-574. <u>https://doi.org/10.1017/s1049023x19005</u>
- Sukontawaree, N., Poonputta, A & Prasitno, A. (2022). Development of Problem-Solving Abilitiesin Science by Inquiry-Based Learning with Cooperative Lesrning for Grade 4 Students, *Journal of Educational Issues*, 8(2). <u>https://doi.org/10.5296/jei.v8i2.20418</u>
- Sumner, A. (2006). What is Development Studies? Development in Practice, 16(6) 644-650. https://doi:10.1080/09614520600958363
- Swami, V. (2016). Re: What is Intact Sampling? <u>https://www.researchgate.net/post/what-is-</u> intactsampling/57206f0e615e279d405e376f/citation/download
- Thomas, L. (2024). Quasi-Experimental Design, definition, types and examples. https://doi.10.1002/978/118785317
- Thompson, D. M. & Wesolowski, B. (2018). The SAGE Encyclopedia of Educational Research, Measurement, and Evaluation: Variance. Thousand Oaks: SAGE Publications. <u>https://dx.doi.org/10.4135/9781506326139.n737</u>
- Thorbecke, E. (2006). The Evolution of the Development Doctrine, 1950-2005. World Institute of Economic Research. <u>https://ideas.repec.org/p/unu/wpaper/rp2006-155.html</u>

- Valbuena, J. G. (2016). Essentialism: As a Philosophy and as a philosophy of Education. Thesis at Philippine Normal University. <u>https://www.scribd.com/document/476941993/Essentialism-As-a-Philosophy-and-As-a-Ph-pdf</u>
- Verma, S. (2020). Music Treatment on Patient on Agitation Level in Early and Middle Stage Alzheimer: Two-way Analysis of Variance, *International Journal of Applied Research*. 6(9), 1-4. https://doi.10.22271/allresearch.2020.v6.i9a.7048
- Virag, J. (2024). The Role of Confidence Level in Statistical Analysis. https://www.statsig.com/blog/confidence-levels-in-statistical-analysis
- Vygotsky, L. S. (1978). *Mind in Society: The Development of Higher Psychological Processes. London:* Harvard University Press
- Wale, B. D. & Bishaw, K, S. (2020). Effects of Using Inquiry-Based Learning on EFL Students' Critical Thinking Skills. Asian Journal. Second Foreign Language Education, 5(9). <u>https://doi.org?10.1186?s40862-020-00090-2</u>
- Wang,Y.(2015).FreireonFreedomofEducation.https://scholarblogs.emory.edu?basicproblems0021/2015/03/freire-on-freedom-of-education/

Williams, K. (2024). Features, Types and Use Cases. https://surveysparrow.com

- Wong, T. k., Tao, X. & Konishi C. (2018). The Teacher Support in Learning: Instrumental and Appraisal Support in Relation to Math Achievement. *Issues in Educational Research*, 28 (1) 202-2019
- Wood, D., Bruner, J. S. & Ross, G. (1796). The Role of Tutoring in Problem-Solving. Child Psychology and Psychiatry and Allied Disciplines, 17 (2), 89-100. <u>https://doi.org/10.1111/j.1469-7610.1976.tt0038.x</u>

Appendices

Appendix 1 Observation checklist

CLASSROOM OBSERVATION CHECKLIST

This checklist aims to assess learners' behaviour during the process of guided inquiry to develop problemsolving skills in the Development Studies classroom. The checklist has been designed with the five phases of Inquiry-Based Learning according to Pedaste et al. (2015).

~ .				<i></i>			
Decorvo loarnore	' hohaviour at	1d mark tho r	<i>~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~</i>	"vøs/no"	according f	h the relevan	t critoria
Observe icumers	ocnuvioui ui	<i>iu mun n me</i> i	csponses,	yesino	uccoraing i	o me recevan	

No	Criteria	Response				
•		s				
Α	Orientation phase: assessment of insightful question	Yes	No			
	posing to identify the problem of investigation.					
1.	Learners demonstrate interest by asking relevant questions					
	to the current investigation.					
2.	Learners can identify the problem of investigation during the					
	process of question posing.					
3.	Learners often ask questions during the process of learning.					
4.	Learners can relate questions to real-world scenarios or					
	problem-solving tasks.					
В	Conceptualisation phase: learners' ability to set goals and					
	procedures to be followed in finding solutions to the					
	problem.					
5.	Learners can identify important objectives to achieve.					
6.	Learners can collaborate and share opinions about the					
	objectives to be achieved.					
7.	Learners can formulate a procedure on how to respond to the					
	identified problem.					

8.	Learners identify the resources or materials that will be	
	needed.	
С	Investigation phase: learners' ability to explore solutions to	
	the identified problem.	
9.	Learners demonstrate curiosity and persistence in seeking	
	relevant information.	
10.	Learners seek answers on their own before asking help from	
	the teacher.	
11.	Learners are able to access resources independently.	
12.	Learners actively collaborate with their peers by sharing	
	ideas to explore solutions.	
D	Conclusion phase: the provision of solutions to the	
	identified problem.	
13.	Learners are able to evaluate the identified solutions to	
	choose the most appropriate.	
14.	Learners are able to place in order of importance the	
	identified solutions.	
15.	Learners shows strong critical thinking skills by making	
	logical connections on the identified solutions.	
16.	Learners provide relevant examples of the identified	
	solutions.	
Ε	Discussion phase: learners demonstrate problem-solving	
	skills, reflect on their process of inquiry and identify areas	
	of improvement.	
17.	Learners evaluates their problem-solving process.	
18.	Learners reflect on their behaviour during inquiry process.	
19.	Learners provide a summary of the identified solutions to	
	demonstrate the development of problem-solving skills.	
20.	Learners reflect on their strengths and weaknesses during the	
	lesson for future improvement.	

Appendix 2 Likert scale

LIKERT SCALE

Aim: To assess learners' satisfaction on the use of guided inquiry-based learning to enhance problemsolving.

NO.	State your level of	Strongly	Disagree	Neutral	Agree	Strongly
	satisfaction regarding	disagree				agree
	underlisted factors					
1.	Guided inquiry approach					
	helped me to grasp difficult					
	concepts more easily.					
2.	With guided inquiry, I was					
	able to apply solutions to real					
	world problems.					
3.	I found it interesting to be					
	given an opportunity to ask					
	questions in class.					
4.	I felt overwhelmed and					
	confused during the process of					
	guided inquiry.					
5.	Guided inquiry encouraged					
	collaboration among us in					
	classroom which helped us to					
	provide solutions to problems					
	easier.					
6.	Collaboration with my					
	classmates challenged me as					
	others had differing opinions					
	which confused me.					
7.	I enjoyed being actively					
	involved in the classroom as I					

	had better understand of the			
	topics being taught.			
8.	In guided inquiry class, I was			
	unable to choose best solutions			
	to identified problems.			
9.	I like being continually			
	supported in the process of			
	learning.			
10.	I was able to identify solutions			
	to the problems independently			
	in guided inquiry-based			
	classroom.			
Appendix 3 Pre-test, Post-test

GRAD	DE 11	DEVELOPMENT STUDIES	MARKS (20)	
GEND	DER:			
AGE:				
1.	Define the folle Population:	owing terms in relation to developm	ent:	
	Overpopulation	n:		
	Infant mortality	y:		
				(3)
2.	Assess the imp	pact of overpopulation on Developm	ent in Lesotho	
	(4)			
	(4)			
3.	Analyse ways	in which infant mortality rate can be	e reduced in Lesotho	

_____(4) 4. Define the term urbanisation _____ (1) 5. Examine the impact of urbanisation in both rural and urban areas _____(4) 6. Describe the possible solutions to the problems brought by urbanisation in both rural and urban areas

_____(4)

Appendix 4 Introductory letter

The National University of Lesotho

Telephone: +266 22340601/3631P.O. Roma 180



Lesotho

FACULTY OF EDUCATION

Date 15thApril, 2024

TO WHOM IT MAY CONCERN

Dear Sir/Madam,

A Letter of introduction to undertake research

This letter serves to introduce Ms. Puseletso Mathaha, who is a Master of Arts in

Education (M.A. ED) student in the Faculty of Education at the National University of

Lesotho (NUL). The student is undertaking research on Development Studies Inquiry-Based Learning approach and problem-solving skills. The study involves classroom-based experimental research for a period not exceeding one month.

Kindly accord him the necessary assistance and support in this important activity.

Your cooperation is highly appreciated.

Yours sincerely,



Dr. Makhulu Makumane (HOD)

Appendix 5 Informed consent

INFORMED CONSENT

Dear parents or guardian

My name is Puseletso Mathaha. I am a Master's candidate at the National University of Lesotho under the Department of Language and Social Education. I am conducting research aimed at exploring the impact of guided Inquiry-based Learning on enhancing problem-solving skills among grade 11 DS learners. In conducting this study, I am hoping to significantly contribute to your children's educational journey if you agree or allow them to participate. Participation shall involve group discussions, hands-on activities, exploration of the physical environment and many more.

All information collected in this study will be kept confidential and stored securely. Again, your child's identity will not be included in any reports or publications regarding this study. Equally important, participation in this study is voluntary. Therefore, you are free to withdraw your child at any time if you find a need to do so.

For any questions, feel free to contact me at <u>puseletsomathahaanna@gmail.com</u> or 51936063

PLEASE READ THE SECTION BELOW AND PROVIDE YOUR SIGNATURE

I have read and understood the above information regarding my child's participation in the study mentioned above.

Appendix 6 Turnitin Report

Page 2 of 125 - Integrity Overview

Submission ID trn:oid:::1:2986782228

14% Overall Similarity

The combined total of all matches, including overlapping sources, for each database.

Filtered from the Report

- Bibliography
- Quoted Text

Match Groups

Top Sources

361Not Cited or Quoted 11%
 Matches with neither in-text citation nor quotation marks

Matches with neither in-text citation nor quotation marks 104Missing Quotations 3%

 7%
 Image: Publications

 1%
 Submitted works (Student Papers)

12%
(Internet sources

0 Missing Citation 0% Matches that have quotation marks, but no in-text citation

O Cited and Quoted 0%
 Matches with in-text citation present, but no quotation marks

Matches that are still very similar to source material